

Review



Survey of Cooperative Advanced Driver Assistance Systems: From a Holistic and Systemic Vision

Juan Felipe González-Saavedra ¹, Miguel Figueroa ¹, Sandra Céspedes ² and Samuel Montejo-Sánchez ³,*

- ¹ Department of Electrical Engineering, Universidad de Concepción, Concepción 4070386, Chile; juanfegonzalez@udec.cl (J.F.G.-S.); mifiguer@udec.cl (M.F.)
- ² Department of Computer Science and Software Engineering, Concordia University, Montreal, QC H3G 1M8, Canada; sandra.cespedes@concordia.ca
- ³ Programa Institucional de Fomento a la Investigación, Desarrollo e Innovación, Universidad Tecnológica Metropolitana, Santiago 8940577, Chile
- * Correspondence: smontejo@utem.cl

Abstract: The design of cooperative advanced driver assistance systems (C-ADAS) involves a holistic and systemic vision that considers the bidirectional interaction among three main elements: the driver, the vehicle, and the surrounding environment. The evolution of these systems reflects this need. In this work, we present a survey of C-ADAS and describe a conceptual architecture that includes the driver, vehicle, and environment and their bidirectional interactions. We address the remote operation of this C-ADAS based on the Internet of vehicles (IoV) paradigm, as well as the involved enabling technologies. We describe the state of the art and the research challenges present in the development of C-ADAS. Finally, to quantify the performance of C-ADAS, we describe the principal evaluation mechanisms and performance metrics employed in these systems.

Keywords: cooperative advanced driver assistance systems; road safety system; holistic; systemic; Internet of vehicles; natural perception; assisted perception; human–machine interface

1. Introduction

The history of research on assistance systems for driving safety is almost as old as the rise of the automobile industry itself. One of these first systems was implemented as standard equipment by Volvo in 1959, when it began to install the seat belt with a three-point system in its vehicles, which was patented in 1962 [1] and later became the universal standard used by commercial vehicles today. Different road safety mechanisms, also known as advanced driver assistance systems (ADAS), have been incorporated over the years in the development of the automotive industry to provide a higher level of road safety to traditional transport systems. These mechanisms constitute technological implementations of concepts that the Electronic and Telecommunications Systems Institute (ETSI) has subsequently defined as road safety applications, which can be grouped into primary road safety applications (equivalent to the so-called active road safety systems) and secondary and tertiary road safety applications (equivalent to so-called passive road safety systems) [2]. The former focuses on collision avoidance as its primary mission, while the latter aims to lessen the severity of potential injuries to vehicle passengers or vulnerable road users (VRU) after the collision. Specific bumpers designed to keep pedestrians safe, airbags, and seat belts are some of the well-known examples of passive safety mechanisms. Others such as the anti-lock braking system (ABS), lane change assistance (LCA), or forward collision warning (FCW) are examples of active safety systems.

Thanks to the technological development achieved in the electronics area, the automotive industry evolved from traditional mechanical control to electronic control of the main internal functions of the vehicle, fundamentally related to acceleration, braking, gear change, suspension, and ignition/alternator integration [3,4]. The principal elements of this



Citation: González-Saavedra, J.F.; Figueroa, M.; Céspedes, S.; Montejo-Sánchez, S. Survey of Cooperative Advanced Driver Assistance Systems: From a Holistic and Systemic Vision. *Sensors* **2022**, *22*, 3040. https://doi.org/10.3390/ s22083040

Academic Editors: Ondrej Stopka, Marek Jaśkiewicz and Milos Poliak

Received: 28 February 2022 Accepted: 11 April 2022 Published: 15 April 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). evolution are the electronic control units (ECU) and the management of the information flow between ECUs, for which bus communication protocols, such as the controller area network (CAN) bus [5,6] or the local interconnect network (LIN), are employed. These developments allowed the implementation of safety systems such as the ABS, the electronic stability control (ESC), or electronically assisted steering (EAS). The introduction of on-board sensors such as radars, ultrasonic, cameras, light detection and ranging (LIDAR), the global positioning system (GPS), and the inertial measurement unit (IMU), enables the development of driving assistance systems more oriented to the sensing and interaction of the vehicle with the surrounding environment to help the driver in the driving process [7]. Initially, these safety systems were oriented to the longitudinal control of the movement of the vehicle, such is the case of the adaptive cruise control (ACC) system, FCW, and autonomous emergency braking (AEB) [8]. Later, lateral control-oriented systems emerged, such as the lane departure warning (LDW) [9], blind-spot warning (BSW) [10], lane-keep assistance (LKA) [11], or LCA [12]. Although these sensor technologies represented a benefit for the surrounding awareness knowledge of the vehicle in terms of road safety, their short-range and line-of-sight obstruction problems limit their ability to sense the environment. A new generation of ADAS, named cooperative advanced driver assistance systems (C-ADAS) [13,14], has started to be developed, which also incorporates wireless communication devices. The main distinguishing factor between the C-ADAS and ADAS is the visibility of the system. C-ADAS can have knowledge of the traffic situation miles ahead, while conventional systems have knowledge of a few meters around the vehicle. The extended visibility is possible not only due to vehicle-to-vehicle (V2V) communication but also to vehicle-to-infrastructure (V2I) and vehicle-to-everything (V2X) communication [15].

Road safety is an area in which internal and external elements converge and interact. Figure 1 shows the main elements involved in a road safety system (RSS), the environment, the driver, and the vehicle. The limitations in the design of ADAS by not considering the presence of all these elements, as well as the interaction among them, degrades the level of road safety and diminishes the effectiveness of these systems.



Figure 1. Main elements involved in a road safety system: environment, vehicle, and driver. Interactions between these elements: vehicle–environment (V–E) interaction, driver–environment (D–E) interaction and driver–vehicle (D–V) interaction. At the center is reflected the holistic relationship between elements and the opportunity in the C-ADAS design.

An important aspect for the proper functioning of C-ADAS is the performance of the vehicular communication networks, which must be able to guarantee the timely exchange of sensitive information for road safety. Vehicular ad hoc networks (VANETs) [16] are powered by vehicular communications and further provide an enhancement in driving experience by improving security, infotainment, and robustness. Many researchers have contributed and explored this concept, but due to many security and privacy-related issues, the implementation stage has not matured enough. Only in the last few years

has there been a growing commercial interest from the automotive industry in the area of vehicular communications. A considerable volume of research in this field has led to a move from conventional VANETs to the Internet of vehicles (IoV) [17]. The fusion between the traditional concepts of VANETs and the most recent concepts linked to the Internet of Things (IoT) has promoted a very tempting research area today. The IoV concept materializes this integration process between VANETs, IoT, and mobile computing. In this sense, the so-called intelligent vehicles are defined, which are equipped with Internet connection devices to collaborate with each other through data exchange and can even interact with other road users, such as pedestrians or roadside units, through the exchange of information about the road environment. Under this concept, a traffic management system can be established based on communication and cooperation between the vehicles, the road infrastructure, and the rest of the actors present in the road environment. On-board sensors and processors with remote connection capacity are required in each element of the network, where the information exchange must be managed efficiently to guarantee that the system is useful and capable of providing a road safety benefit.

In this work, we present a review of the previous works that have addressed the interactions between the principal elements of the RSS (i.e., driver, environment, and vehicle) and the challenges that currently persist in each area. Then, we present a proposal for the general vision of the C-ADAS design. Note that, unlike the previous studies, in this work we intend to describe a holistic and systemic architecture while highlighting the main works that have already been carried out in each type of interaction. Besides, we describe the elements that we consider fundamental for proposing a C-ADAS design that seeks to bring together all the technological advantages from the point of view of algorithms, communications, and sensing capabilities. The main contributions of this work are: (i) To propose a novel C-ADAS architecture with a holistic and systemic approach that considers the main elements of the RSS (driver, environment, and vehicle) and their interactions. To the best of our knowledge, it is the only survey that addresses the interactions between each of these elements from a bidirectional approach, considering the relevance of information exchange, highlighting cooperative solutions and the need for intelligent techniques in decision making; (ii) To present a study on the main evaluation metrics and mechanisms used for the validation of C-ADAS, which has been little addressed in the previously published surveys.

The rest of this paper is organized as follows: In Section 2, we describe the main characteristics and limitations of previous surveys on ADAS and the current challenges in this subject, while in Section 3, we propose a novel C-ADAS architecture from a holistic and systemic perspective. Sections 4–6 review the details associated with the interactions of the three principal elements (driver, environment, and vehicle) of the RSS, mentioning its actual state and challenges. More specifically, Section 4 presents an overview of the interactions between vehicle and environment, defined as an assisted perception module, Section 5 details the interaction between driver and vehicle, mainly described through the HMI module, and Section 6 presents the interactions between driver and environment, defined as the natural perception module. Section 7 presents the most used metrics and mechanisms in the evaluation and validation of C-ADAS systems, mentioning some of the main challenges in this area that, according to our vision, hinder the implementation of a C-ADAS system, such as the one proposed in this article. Finally, we conclude the paper in Section 8 and consider some directions for future work.

2. Previous ADAS Surveys and Current Challenges

There are several published surveys on ADAS in connected and automated vehicles (CAV) with different design perspectives. Despite the latent needs and challenges of the integration and bidirectional interactions of the three subsystems analyzed, to the best of our knowledge, these aspects have not been addressed in the scientific literature. However, some surveys have stood out for the rigor and depth with which they address some of these

elements. In this section, we present the aspects in which each one of them stands out in order to later point out the shortcomings and discuss the current challenges in the subject.

- Pathan et al. [18] provide a review of the proposed techniques to implement C-ADAS and intelligent traffic management systems, comparing pros and cons, and also look at the practically feasible features;
- Hasenjäger et al. [19] provide a review of the personalization for ADAS and propose a general conceptual framework of personalized ADAS and the human–machine interface (HMI), which can be expected to continuously adapt in interaction with the driver;
- Martinez et al. [20] provide a survey on driving-style characterization and recognition, revising several algorithms with emphasis on machine learning approaches;
- Xing et al. [21] present an overview of the driver intention inference, which mainly focuses on the lane change intention on highways;
- Bila et al. [22] provide an overview of information and communication technologiesbased support and assistance services for the safety of future connected vehicles, given from the perspective of vehicle detection, road detection, lane detection, pedestrian detection, drowsiness detection, and collision avoidance;
- Siegel et al. [23] summarize the state of the art in connected vehicles, reviewing the architectures, enabling technologies, applications, and development areas;
- Wang et al. [24] focus on heterogeneous multi-sensors fusion technologies, including radar, camera, LIDAR, ultrasonic, GPS, IMU, and V2X communication, analyzing the necessity of fusion strategies because of the limitations of sensors;
- Martí et al. [25] propose an overview of existing and upcoming sensor technologies applied to common perception tasks for ADAS and automated driving. Specifically, they focus on artificial vision, radar, and LIDAR technologies of exteroceptive sensors applied in tasks as (i) automatic traffic-sign detection and recognition, (ii) perception of the environment, and (iii) vehicles, pedestrians, and other obstacles detection;
- Kaiser et al. [26] carry out an author-centric literature review to illustrate the opportunities in using smartphones to detect driver distraction. The authors have reviewed several papers and summarized their application cases, smartphone sensor data used, methods, and results;
- Arumugam et al. [27] survey driver behavior analysis based on the use of big data. Works related to monitoring driving patterns and fatigue, detecting drowsiness and driver distraction, are discussed;
- Zhang et al. [28], investigate mean take-over times from 129 studies with SAE (Society
 of Automotive Engineers) level two automation or higher. How quickly drivers takeover control of the vehicle in response to a critical event or a take-over request is an
 important question in automated driving research;
- Sarker et al. [29] review the principal aspects of the sensing and communications technologies, human factors, and controller aspects for information-aware CAV.

Table 1 summarizes these works and shows that none of them have described the ADAS architecture from a holistic and systemic perspective; evaluation mechanisms are hardly mentioned and only one of them has addressed the interaction between the three subsystems, but not bidirectionally in all cases.

Table 1. Analysis of the main characteristics and limitations of other published surveys and this work in terms of architecture, system evaluation, and interactions between elements, considering D–V interaction, D–E interaction, and V–E interaction; specifying whether this relationship is bidirectional (<->) or unidirectional (->). Cases where these topics are not addressed (N/A) are also highlighted.

Articles	Architecture	D-V Interactio	n D–E Interaction V	/-E Interaction	System Evaluation
[18]	N/A	D <-> V	N/A	V <-> E	N/A
[19]	Modular	D <-> V	N/A	N/A	Mechanisms
[20]	N/A	D <-> V	N/A	E -> V	N/A
[21]	Modular and relational	D <-> V	D -> E , E -> D	E -> V	Mechanisms and metrics
[22]	N/A	N/A	D -> E , E -> D	V <-> E	N/A
[23,24]	N/A	N/A	N/A	V <-> E	N/A
[25]	N/A	N/A	N/A	E -> V	N/A
[26]	N/A	N/A	D -> E , E -> D	N/A	N/A
[27,28]	N/A	D <-> V	D -> E , E -> D	N/A	N/A
[29]	Modular	D <-> V	N/A	V <-> E	N/A
This work Modular, holistic, and systemi		c D <-> V	D <-> E	V <-> E	Mechanisms and metrics

Current Challenges

A holistic approach of the system in the interactions between vehicle–environment, vehicle–driver, and driver–environment, which are essential for improving road safety, has not been sufficiently addressed in the literature consulted. Holistic is an adjective that indicates that something is relative or belongs to holism. Holism is a concept created by Jan Christiaan Smuts [30], which he describes as "the tendency of nature to use creative evolution to form a whole that is greater than the sum of its parts". In general terms, holistic indicates that a system and its properties are analyzed as a whole, in a global and integrated way, through the multiple interactions that characterize them. Holism assumes that all the properties of a system cannot be determined or explained as the sum of its elements, highlighting the importance of the inter-dependence of those elements. The term systemic is used in the literature with a similar meaning to the term holistic; however, in this work when referring to a systemic perspective we focus mainly on the structural aspect of the system, in the analysis of each of the elements that compose it, and in the bidirectional study of the interactions that occur among them.

In this sense, the design of current ADAS has a variety of potential difficulties: reflecting the effects of all kinds of traffic factors on driving safety; describing the interactions between the characteristics of the driver's behavior, the state of the vehicle, and the road environment; or providing an accurate basis for vehicle control. Existing systems that assess driving safety may not work properly if they consider only a limited number of factors and their interactions. Driving is a complex decision-making process due to the intricate relationships between the main elements (vehicle, environment, and driver) and the dynamic nature of these elements. In the driver–vehicle–road closed-loop system, the driver is a crucial component, with unique driving characteristics that vary from driver to driver or even for the same driver under different conditions or on different days.

The directionality of the interactions between vehicle, environment, and driver is very relevant. For example, consider a situation where a driver approaches an intersection, in which the traffic light just switched to red. In this scenario, the environment is issuing a notification about the state of road safety, representing a form of interaction in the direction environment -> driver ($E \rightarrow D$) and environment -> vehicle ($E \rightarrow V$). This notification can be captured correctly or not by the driver. The vehicle, through its on-board sensors, could detect the red light and then calculate the distances between the elements of the

environment, e.g., the predecessor vehicles and the intersection. Besides, the surrounding situation is transmitted to other road users through messages corresponding to the V -> E interaction. On the other hand, the vehicle can sense the driver's reaction to surrounding stimuli by monitoring the driver through on-board devices, which is a form of $D \rightarrow V$ interaction. The system alerts the driver through the V -> D interaction if their reaction is not as expected and communicates to other road users the potential risk, as a V -> E interaction form. Additionally, the vehicle captures the driver's reaction to the C-ADAS alert notice, such as a D -> V interaction form. Consequently, the C-ADAS takes control of the vehicle to guarantee road safety if the driver's reaction is not as expected, which is a form of $V \rightarrow E$ interaction. In this way, it is observed how the C-ADAS system is responsible for assisting the driver in a timely manner on road safety risks, guaranteeing the correct closure of the cycle started with the notification of the environment to the driver on the red light (the E -> D interaction), finishing with a correct response in relation to the reaction that the driver should have in this situation ($D \rightarrow E$ interaction), preserving road safety with or without the intervention of the C-ADAS at different levels. This shows how C-ADAS can mitigate a poor reaction on the part of the driver, resulting from a wrong interpretation of the situation, or the driver's inability to react.

The RSS consists of a driver subsystem, a vehicle subsystem, and a road environment subsystem. These three subsystems are essential to ensure the safe driving of automobiles. The "gaps" in the interactions between the three subsystems are causes of degradation in road safety. A fundamental element to fill the gaps in these interactions is the design of ADAS from a holistic approach that considers the presence and interaction between these different elements as a single system. Recently, there have been situations in which the challenges faced by the automation of vehicle systems are emphasized, demonstrating how the non-consideration of any of these elements can produce unfavorable results. Examples of this are some tragic accidents that occurred in Tesla vehicles [31–33], highlighting the serious life-or-death consequences associated with the failures or miscalculations that can occur with vehicular systems. A Model X issued several audible and visual alerts before crashing into a concrete wall without the driver putting their hands on the wheel [31]. A preliminary report from the National Transportation Safety Council (NTSC) found that the driver of the vehicle activated autopilot, which provides automatic driving functions, 10 s before impact [32]. The document also indicates that the automatic system did not detect the driver's hands on the wheel in the last eight seconds and that it did not perform evasive maneuvers to avoid colliding with a truck. In reference [33], the Tesla driver acknowledged that he was watching a movie on their mobile phone at the time of the accident. However, the vehicle never got to take control of the situation, necessary in view of the evident state of distraction of the driver. In this particular case, perhaps the first and foremost thing is to slow down the rush to "remove the human being" from the equation. Officially, Tesla's autopilot was meant to aid the driver, not replace them. Human beings remain essential to driving and should continue to be so for some time until the technology matures. It is crucial to understand the surrounding cars with respect to the road context and to interact with them harmoniously for the success of the autonomous cars used in mixed urban traffic [34].

Similar examples can be evidenced by excluding any of the other two elements (vehicle and environment) of the system in the design of ADAS. The increase in road safety has motivated progress in the area of safety systems. Depending on the degree of automation of these, their action can be limited to driver assistance tasks or to be even of a more autonomous nature: acting directly on the components that modify the response of the vehicle, such as the acceleration and braking pedals and/or the power supply and steering control. The complex interactions between these characteristics and the actions of drivers have led to numerous investigations on the human factors involved in motor vehicle accidents. Some of these factors are demographics, distraction, experience, fatigue, alcohol, stress, the tendency to risk behavior, and decision making. At present, ADAS are a promising field of research in order to improve road safety.

While it is true that current projections on the levels of autonomy of modern vehicles tend towards a fully autonomous design, the fifth level according to SAE [35], the process of the mass inclusion on the roads of fully autonomous vehicles and their coexistence with human-driven vehicles is not expected to end in the short term [36]. Moreover, it should be noted that the very process of the development and improvement of the artificial intelligence systems that control autonomous vehicles requires for the continuous learning of the interactions between other drivers and road users, which is, in essence, learning to deal with human behavior. In the evolution of the levels of autonomy of vehicles, it is necessary to guarantee road safety in order to preserve lives and limit economic losses through accident avoidance. To this end, the implementation and development of an effective ADAS that accompanies and assists the evolution and maturity of autonomous driving is essential.

3. General Description of the Ideal Architecture of a C-ADAS

In this section, we describe our C-ADAS architecture proposal from a holistic and systemic vision. The systemic vision considers the relevance of the three main elements that make up the RSS (driver, environment, and vehicle). For its part, the holistic vision also reflects the relevance of considering the dependence and interrelationship that exists between each of these elements as essential requirements for the functioning of the system as a whole. Additionally, we establish a comparative analysis between this proposal and the architectures proposed in the previous reviews, describing its main characteristics and limitations. Figure 2 presents our C-ADAS architecture proposal.



C-ADAS

Figure 2. Conceptual architecture of a cooperative advanced driver assistance system (C-ADAS) from the holistic and systemic vision. The red, green, and blue lines indicate the local operation of the principal module (inside the vehicle), while the cyan lines refer to the remote operation of the principal module through the vehicular network under the IoV concept.

This architecture considers the presence of the three main elements of an RSS as independent and closely interrelated subsystems, which influence and determine the state of road safety. The driver subsystem includes everything related to the actions and reactions of the drivers during the driving process, conditioned by their physical and psychological characteristics, such as age, gender, mental state, driving skills, experience, and driving profile. The vehicle subsystem considers aspects related to the electromechanical characteristics of the vehicle, braking behavior, acceleration, and sensing and communication technologies embedded on board, among others. Finally, the environment subsystem takes into account the rest of the elements outside the vehicle and the driver; among them, we can mention the elements of the road structure, the physical conditions of the road, the weather conditions, as well as other vehicles, pedestrians, animals, or obstacles on the road. We propose an integral communication system that seeks to leverage the advantages offered by the IoV concept, using data integration and the capacity for centralized decision making. Note that taking C-V2X communications into consideration implies a high degree of redundancy so as not to lose communication and to have several sources through which redundant data may arrive. Furthermore, the heterogeneity of networks and technologies may generate significant variations in the perceived delay and jitter among different communication paths, which become very sensitive issues for road safety applications. That is why a C-ADAS must conceive minimum delay times for the security modules, which could imply that in some cases the information that causes an action could be only conditioned by messages received from neighboring vehicles in the vicinity of the ego vehicle. In [37], an in-depth discussion of the control aspects of CAV systems is carried out, highlighting the main challenges generated by the existence of different information flow topologies, mainly focused on string stability, communication issues, and dynamics heterogeneity. These aspects must be taken into consideration for a practical implementation of a C-ADAS, such as the one proposed here, since the heterogeneity in the network topology and the high dynamism of vehicular networks can impose additional communication challenges. However, we are committed to the idea of conceptualizing the use of on-board communication devices with C-V2X functionality, framed within the IoV concept, given that current and future developments point to the convergence of technologies. In this sense, 5G is seen as a key technology to support the development of future vehicular networks, and presents low-latency values in communications, which will undoubtedly help to compensate for the delays associated with the processing and control of communication schemes.

This architecture includes three modules to capture the data from the bidirectional interaction between the three main elements of the RSS, these are the natural perception module, the assisted perception module, and the human-machine interface module. Additionally, a main module is included in an upper layer that receives, processes, and stores the collected information by the lower modules to make decisions about the state of road safety. This is a modular architecture where the main module can be implemented both locally, inside the vehicle, and remotely, in a centralized management station within the environment of intelligent transportation systems (ITS) under the IoV concept. In the latter case, the entire flow of information from the three data capture modules would be shared through the V2X communication devices on board the vehicle. From the review of the state of the art in articles of enabling technologies, we can divide them into sensor technologies, wearable technologies, and communication technologies, which are related to each of the data capture modules and their connections with the main module. Although the most researched enabling technologies are those that intervene in the vehicle-environment interaction, it is also important to highlight those that are part of the vehicle-driver and driver-environment interactions, since all of them together allow for a closing of the cycle in a holistic and systematic design of cooperative ADAS, where the three main elements of an RSS are present: driver, vehicle, and environment. In this sense, we have tried to list below the most relevant technologies (described repeatedly in many of the articles consulted), but at the same time to reflect, through an integrating spirit, the technologies present in each

of the three areas of interaction between the three main elements of the RSS. For a greater level of detail, the reader can consult these technologies in references [23,25,27,29,38,39]. (i) Sensor technologies:

- (1) Sensor technologies:
- Radar systems are classified into: (i) short-range radars (SRR), which have a detection range of up to 20 m, are based on a single antenna, and are not capable of detecting angular information; (ii) long-range radars (LRR), which have a range of up to 150 m and an angular resolution of up to 2 degrees [40];
- Cameras embedded on vehicles are of two fundamental types: (i) stereo cameras used to obtain a wide panoramic vision in conditions of good visibility; (ii) infrared cameras used in situations of reduced visibility at night or in the presence of adverse weather conditions [41];
- Light detection and ranging (LIDAR) uses laser signals to determine the relative distance of nearby objects from a vehicle. Laser signals are emitted and their respective echo signals are received to calculate these distances, with detection ranges between 10 and 200 m [42];
- Acoustic sensors use an operating principle similar to that of radars and LIDAR, but they use high-frequency sound waves (ultrasonic) to determine the distance of an object to the vehicle [43].

(ii) Wearable technologies (electronic devices designed to attach to the user's body classification depends largely on the device's functional properties):

- Smartwatches are electronic devices with functionalities such as GPS, fitness/health monitoring, and waterproof operation [44];
- Wearable cameras are much more flexible and mobile than conventional cameras, as they focus on a first-person view and are often attached to eyeglasses, helmets, and caps [45];
- Smart eyewear are used to provide information, notifications, and a three-dimensional view through optical head-mounted displays (OHMDs), heads-up displays (HUDs), virtual reality (VR), augmented reality (AR), and/or mixed reality (MR) [45];
- Fitness trackers are placed on different parts of the body to monitor the physical state of the individual during the performance of daily activities or exercise routines. Their measurements include parameters such as speed, heart rate, calories released, and number of steps [46];
- Smart clothing—usually shoes, hats, clothing, and helmets—incorporate cameras and sensors to monitor body signals and adapt their characteristics to the individual's state [47];
- Wearable medical devices made up of one or several biosensors are used to monitor the physiological activity of the individual for the purposes of the prevention, diagnosis, and early treatment of diseases and health status abnormalities by measuring temperature, heart rate, blood pressure, and glucose level, or by performing electrocardiography (ECG), electroencephalography (EEG), or electromyography (EMG) [48].
 (iii) Communication technologies:
- Dedicated short-range communication (DSRC) is a technology of fully integrated vehicular networking, implemented over the 75 MHz bandwidth (5.85–5.925 GHz) assigned for the Federal Communications Commission (FCC). The architecture and the services to enable these secure V2V and V2I communications in the Wireless Access Vehicular Environment (WAVE) are provided by the IEEE 802.11p and IEEE 1609 protocol suites [49];
- Light fidelity (Li-Fi) uses wireless communications in the visible-light band for data transmission by encoding the flashing states of light-emission diodes (LEDs) [50];
- LTE Advanced Pro (LTE-A) is part of the evolution of LTE networks in version 14 to ensure that V2X service requirements are supported by the LTE transport network. Different V2X application scenarios are defined, including V2V, V2I, vehicle to pedestrian (V2P), and vehicle to network (V2N) [51];

- The "5G/IMT-2020 Standing Committee" reports the standardization process of fifthgeneration wireless communication technologies (5G) [52]. This technology is expected to be the future of vehicular communication networks, providing support for lowlatency and ultra-reliable communications (URLLC) scenarios.
- The internal communication networks of a vehicle are composed of electronic control units (ECUs), mechanical and electric sensors, and actuation devices to guarantee the correct vehicle operation. Original equipment manufacturers (OEMs) today design proprietary devices and networks to share data through on-board diagnostic (OBD) hardware. Well-established technologies such as CAN bus, MOST, LIN, and FlexRay are examples of their resilience and flexibility. Other emerging technologies such as the vehicular Ethernet support the growing communication capacity demanded for modern vehicles, and are discussed in reference [53].

The main module is in charge of using the information from the three data capture modules to analyze the state of road safety and make the best decision when it comes to assisting the driver or acting directly on the vehicle's safety systems in case of the nonattention of the driver regarding warning notices to an imminent danger situation, with the aim to minimize the level of risk and traffic accidents. It is also in charge of sharing this information with the other actors in the road environment. The operation of this module begins with a reception information block that receives the data from the three capture modules. Once this data has been processed, models of driver behavior are established and stored, and customized according to the characteristics of the driver. These models allow for estimating the prediction of the longitudinal and lateral movement of the vehicle, information that can be used by different advanced driver assistance systems (ADAS-1 ... ADAS-N blocks) either to provide assistance to the driver or to take direct control of the vehicle in case the driver does not respond adequately and in a timely manner to certain dangerous situations on the road. In this sense, the decision making of the C-ADAS system may entail carrying out the functions of vehicle control, visualization through the HMI, or the transmission of information related to road safety to the rest of the actors on the road or to the centralized management station in an ITS.

Under the IoV concept, different multi-layer architectures have been proposed and different types of interactions between the elements that compose it have been established [17,54]. One of the layers that is commonly addressed is the centralized platform in the cloud (associated with the "intelligent brain" in the IoV architecture). In this, the centralized processing of the road safety data obtained in the lower layers is carried out, as well as the global strategies and the management of road safety alerts, assisted automatic driving, and intelligent navigation, among other functions. Centralized or distributed execution implies differences in the signaling overhead and delay associated with decision making, but the processing delay in the cloud is less than in the computer equipment inside the vehicle. Some critical security applications will demand URLLC-type communications, which will benefit from the growing development of 5G. The analysis of the local data allows to guarantee the control of the main functions of the vehicle in real time, while the remote analysis of the same allows to create added value in the system to improve the reliability, the efficiency, and the performance of the vehicle through the application of communications, computational processing, and distributed information on a large scale. To the extent that connectivity capabilities and computer technologies are developed and latency in the execution of applications is reduced, the use of aggregated data, obtained remotely, becomes more useful in the operation of these road safety applications in real time. Vehicles do have limited computation and storage resources that may not be sufficient for road safety applications that need to process a big amount of data. Since they require big storage and complex computations, for this, vehicle-to-cloud computation helps by providing proficient support to these applications. In our architecture, the main module is flexible in order to operate with some of these functions, and is mainly oriented to the design of advanced driver assistance systems. Another layer that is described in an IoV architecture is the data acquisition layer, whose main function is to collect different types

of data from different sources and to digitize the data to ensure that it can be successfully transmitted and analyzed. In our architecture, this IoV layer is represented by the three data capture modules: the natural perception module, the assisted perception module, and the HMI module, described below.

The natural perception module is responsible for capturing the elements of the D– E interaction during the natural perception process from the driver of the surrounding environment. Through devices implemented inside the vehicle, such as vision cameras, heart-rate-monitoring devices, and ethyl breath detection, the driver's behavior can be analyzed based on the monitoring of their activity on board and their reactions to the dynamics of the environment, which can help infer their behavior and future actions on the road [21].

The assisted perception module is responsible for capturing the elements of V–E interaction through sensors, such as as radars, ultrasonic, GPS, IMU, LIDAR, and cameras, jointly with communications equipment, meaning that the state of the surrounding environment of the vehicle is sensed and its dynamic data are captured for the posterior presentation to the driver as additional information of their natural perception. This module is also responsible for communicating the driver's mobility and status information to the environment while driving, as well as ADAS alerts and warnings and the driver's reaction to them.

The HMI module is responsible for capturing the elements of the D–V interactions during the driving process. It is composed of the human–machine (H-M) visualization devices, such as displays, vision cameras, or instrument panels, and by the H–M control devices, such as the acceleration, brake, and clutch pedals, the steering wheel, switches, panel buttons, and the gear lever, etc. The four principal functions of this module are: (i) sensing the physical actions of the drivers over the direct and indirect devices involved in the driving process, such as the steering wheel, the brake, clutch, and acceleration pedals, the transmission state, lights, and other on-board controls; (ii) displaying information about the vehicle status (from the internal modules of the CAN bus and GPS modules) and the surrounding environment (maps, routes, traffic signs, location of other vehicles, pedestrians, and obstacles on the road); (iii) visualizing alerts for the driver regarding dangerous situations as a product of the analysis of data related to road safety; and (iv) sensing the driver's reactions to the alerts emitted by the assistance system.

Control systems refer to where C-ADAS processing and decision-making takes place, it can be locally in the vehicle itself or remotely, under IoV concept. The modular structure describes the elements that make up the system and their functions. Personalization refers to the degree of mutual adaptation between the system and the driver during the vehicle-driver interaction, it includes the latter's actions during the driving process, as well as its reaction to the system alert notices presented through the HMI interface of the vehicle. Cooperative communication highlights the capacity of the system to communicate to other actors in the road environment about the information regarding the driver's actions and the result of the system's decision-making regarding the state of road safety. The assistance functions basically refer to assisting the driver or taking control of the vehicle's safety systems. It also highlights the system's ability to operate simultaneously with more than one ADAS (LCA, FCW, AEB, BSW, LDA, among others).

However, monitoring the progress of these technologies, as well as the achievements and projections of the main automakers, will favor the implementation of a proposal close to this C-ADAS that efficiently exploits bidirectional interaction between the three subsystems to increase road safety.

Table 2 resumes the analysis of the architectures proposed in the surveys consulted and our architecture proposal, highlighting the control systems, the assistance functions, and the degree of personalization and cooperation, as well as the main elements that compose them, describing also how the interactions between these elements are implemented.

Articles	Control System	Modular Structure	Personalization	Cooperative Communication	Assistance Function
[19]	Local control	System composed of three main modules: ADAS module, personalization module, and HMI module.	Personalization module that continuously adapts the ADAS to the driver's behavior through the HMI interface.	Cooperative communication is not explicitly considered.	Consider vehicle control and multiple driver assistance functions.
[21]	Local control	System composed of six main modules: environment perception module, vehicle dynamic module, driver behavior recognition module, driver intention inference module, lane change decision module and interaction module.	The interaction module models the driver hand and foot dynamics as well as the dynamics of the vehicle–control interface.	Cooperative communication is not explicitly considered.	Consider vehicle control and a single assistance function: lane change intention inference.
[29]	Local and remote control	System composed of three main elements: sensing and communication technologies, human factors, and information-aware autonomous vehicles controllers.	Human factor element: design of a CAV system based on human driver's expectation, and adaptation of the human driver to the designed CAV system.	It considers the exchange of environment information through the sensing and communication modules, but not the alerts and warnings generated for the ADAS system.	Consider vehicle control and multiple driver assistance functions.
This Work	Local and remote control	System composed of three modules that sense the interaction between the three elements of road safety: the natural perception module, assisted perception module, and HMI module. Besides, a principal module that receives, processes, stores, makes decisions, and transmits the safety information.	Continuously adapts the ADAS by monitoring the driver's condition and their physical interactions with the vehicle's driving-control elements, as well as its reaction to the information and alerts issued by the HMI module.	It considers the exchange of information through the communication devices on board the vehicle, both the information obtained by the three sensing modules and the alert or warning information generated by the main module of the system.	Consider vehicle control and multiple driver assistance functions.

Table 2. Analysis of the ADAS architecture proposed in the revised surveys and the architecture proposed in this work.

4. Vehicle-Environment (V-E) Interaction

The analysis of bidirectionality in the study of the interaction between driver and environment (V-E) is relevant to all ADAS. If we analyze, for example, the information coming from the environment, we find that it can be received by the C-ADAS through several ways: (i) directly, through the information captured by the set of sensors and communication devices that are found on board the vehicle ($E \rightarrow V$) or else (ii) indirectly, at first through the information captured by the driver through their sensory elements and later, through the information partially captured by devices on board the vehicle in charge of monitoring the status and behavior of the driver as a reaction to the stimuli that he perceives from the environment ($E \rightarrow D \rightarrow V$). This alternative path of redundancy is relevant to consider, given that sensors and communication devices present limitations and challenges for their optimal operation. This degree of redundancy can only be achieved if the design of the C-ADAS is approached from a holistic and systemic perspective, where the study of directionality in interactions is considered. The works of the state of the art that are described in this section, fundamentally contribute to the first of these two ways (directly).

The main elements associated with the assisted perception module of the surrounding environment are described in this section, as well as the signaling devices, sensors, and communication technologies involved with the operation of this module. On the one hand, this perception "assists" the driver in acquiring information from the environment in which he/she operates. On the other hand, the vehicle, through its signaling and communication devices, communicates to the surrounding environment information on its movement state and also information related to the intention and actions of the driver. In a general sense, the sensors and communication technologies improve the range and precision of the information that the driver can perceive, especially in variables that are difficult to estimate, such as distances, speeds, and the relative accelerations between vehicles and other actors in the environment.

With the use of communications, high-level information can be incorporated with a high predictive degree, which allows for the anticipation of changes in the dynamics of vehicle movement before they materialize and can be perceived by drivers. In the same way, the range of perception coverage can be extended and information can be obtained beyond the local environment of the vehicle, which is associated with scenarios where there are no direct-line-of-sight conditions.

4.1. Actual State

The discussion of this section begins by grouping the works that use vision and infrared cameras to capture the information from the vehicle–environment interaction, a method that is currently widely used in the automotive industry, but with limitations related to adverse weather conditions, lighting deficiencies, and the high computational cost of image processing algorithms, to name a few. Xin et al. [55] propose an intention-conscious model to predict the trajectory based on the estimated lane change intention of neighboring vehicles, using an architecture with two long–short-term memory (LSTM) networks. The first one receives as input sequential data that characterizes the lateral movement of the vehicle to infer the driver's intention to stay in the lane, turn left, or turn right. Once the target lane is detected, this indicator is passed as an input to the second LSTM network, which also receives the sequential data of the longitudinal movement of the vehicle, to finally predict its position. From the view point of the ego vehicle, only the features that can be feasibly measured using on-board sensors, such as LIDAR and radar, are used as input. The database used in this paper is from the next generation simulation (NGSIM) [56].

Deo et al. [57] design an LSTM encoder-decoder model that uses convolutional socialgrouping layers as an enhancement of social-grouping layers for the robust learning of inter-dependencies in vehicle movement. The social grouping is defined by a structure called the social tensor, which groups the LSTM states of all the agents located around the predicted agent. This is done by defining a spatial grid around the agent being predicted and filling the grid with LSTM states based on the spatial configuration of the agents in the scene. The encoder is an LSTM network with shared weights that learns vehicle dynamics based on trajectory histories. The output of the LSTM decoder generates a probability distribution on future movement for six maneuver classes and assigns a probability to each maneuver class. A lot of complementary information can be captured using visual- and map-based cues. For the experiments, they use the publicly available NGSIM database. Deo et al. [58] propose a variation of the architecture designed in [57], using an LSTM network as an intermediate layer to classify and assign a probability to the maneuvers instead of the convolutional social-grouping layer. The results of these analyses evidenced the importance of modeling the movement of adjacent vehicles to predict the future movement of a given vehicle, as well as the importance of detecting and exploiting common vehicle maneuvers for the prediction of future movement. They use the publicly available NGSIM database for the experiments.

Kim et al. [59] present a collision risk assessment algorithm that quantitatively assesses collision risks for a set of local trajectories through the lane-based probabilistic motion prediction of surrounding vehicles. Initially, the target lane probabilities are calculated, representing the probability that a driver will drive or move into each lane, based on lateral position and lateral velocity in curvilinear coordinates. It assumes that the lateral offset of vehicles with respect to the road center-line is measured from a suitable sensor suite, such as a camera, radar, or LIDAR. To estimate the collision probability, the collision risk is assumed as a metric, which is modeled as an exponential distribution, dependent on the time to collision (TTC). The prediction performance of the lane-based probabilistic model is first validated by comparing

the model probabilities from the probabilistic target lane detection algorithm against the maneuver probabilities obtained from real-world traffic data from the NGSIM database.

Hou et al. [60] propose a model of mandatory lane change (maneuver of incorporation into the vehicular flow of a highway) that considers as input variables data associated with the distances and relative speeds between the vehicle that is going to carry out the maneuver and the front and rear vehicles in the lane to which it is intended to enter, in addition to the distance the vehicle has traveled on the merging lane. Bayesian classifiers and decision trees are used to predict the driver's decision to carry out the maneuver or not, determining as the most relevant variable the relative speed between the vehicle carrying out the maneuver and the vehicle in front on the line to which it is intended to enter and, in general, the greater relevance of relative speeds over relative distances. Detailed vehicle trajectory data from the NGSIM database were used for model development (data of U.S. Highway 101) and testing (data of Inter-state 80). Liu et al. [61] develop a deep learning model to evaluate discretionary lane change maneuver decision making. The model is based on deep neural networks and with the exception of the instantaneous states of the subject and the surrounding vehicles, the historical experience of the drivers and the memory effect from vehicle to vehicle are also taken into account for the final evaluation of the maneuvering situation of change of lane, considering the analysis of the time series of trajectory data as part of the historical behavior of drivers. The classifier used is a gated recurrent unit (GRU) neural network, which is a type of RNN. They use the traffic data of the NGSIM database to train and test the model.

Benterki et al. [62] present a system for predicting lane change maneuvers on motorways. These maneuvers are classified into left turn, right turn, and lane keeping, using two machine learning techniques: a support vector machine (SVM) and neural networks. The system also estimates, with a time window in advance, the time in which the lane change maneuvers will take place. The lane change process is subdivided into three stages: preparation of the lane change, active execution of the lane change, and completion of the lane change. Therefore, the system proposes to exploit the changes that occurred during the lane change preparation stage for the premature detection of maneuvers. The real-driving data of the NGSIM database is used for training and testing. Ding et al. [63] propose a method that combines high-level policy anticipation with low-level context reasoning. An LSTM network is used to anticipate the vehicle's driving policy (go ahead, yield, turn left, and turn right) using its sequential historical observations. This policy is used to guide a low-level optimization-based reasoning process. In this reasoning process, cost maps are defined to represent the context information, which are associated with certain characteristics of the road, such as lane geometry, static objects, moving objects, the area enabled for driving, and speed limits. The open-source urban autonomous driving simulator, CAR Learning to Act (CARLA) [64] is adopted to collect the driving data, with the use of a Logitech G29 racing wheel.

Mahjoub et al. [65] propose a stochastic hybrid system with a cumulative relevant history based on GPs. This design is used within the context of model-based communication to jointly model driver/vehicle behavior as a stochastic object and obtain accurate predictive models for mixed driver/vehicle behavior trends in the short and long term (within 0–3 s) of the critical dynamic states of the vehicle, such as its position, speed, and acceleration, within the discrete modes of the system, which are equivalent to the different long-term behaviors (maneuvers) of the driver. The lane change maneuver is selected as a specific long-term driver behavior, and the lateral position of the vehicle is modeled through an available set of already-observed instances. This is done by building a cumulative training history of on-the-go maneuver-specific data from identical or relevant maneuvers observed in the driver's recent driving history, and then feeding this training data to the model inference block, such as your initial training set. To evaluate the proposed method, real trajectory data of 40 lane change maneuvers from the NGSIM database were used. As a recommendation for network situations with a high degree of congestion, where frequent reception of messages is difficult, a model that combines the constant speed model with the

proposed Gaussian regression model would ensure prediction for both the near future (less than one second), as well as for the far future (between one and three seconds).

The following describes the works that use more specific sensors of greater complexity and economic cost, such as radar, LIDAR, GPS, and IMU, to capture the information from the interaction with the vehicle environment. These devices, on the other hand, have disadvantages due to direct-line-of-sight obstruction problems, in the case of LIDAR radars, and in general to unfavorable environmental conditions. Batsch et al. [66] propose a classification model using a Gaussian process (GP) for the problem of detecting the presence or absence of the risk of collision between a vehicle and the vehicle that precedes it, which circulates at a slower speed as a result of being part of a traffic congestion scenario that includes several vehicles. To train the system they use data produced by CarMaker simulation software [67]. The tests are conducted in an automated vehicle equipped with a radar sensor, neglecting the sensor uncertainty in the velocity and aperture angle measurement. Zyner et al. [68] present a method based on recurrent neural networks to predict driver intention by predicting multi-modal trajectories that consider a level of uncertainty. To deal with data sequences of different lengths, sequence-fill techniques are introduced, taking as reference the last known position of the vehicle. The data analyzed contains the lateral and longitudinal position track history, as well as heading and velocity. Park et al. [69] employ the use of LSTM networks to predict the future trajectory of surrounding vehicles based on a history of their past trajectory, formulating the vehicle trajectory prediction task as a multi-class sequential classification problem. For the evaluation of the system, real-vehicle trajectory data from a highway environment was employed. To capture the vehicle-environment interaction data, the test vehicle used a radar sensors and the IMU sensors.

Next, the works that use, together with the use of cameras, the incorporation of sensors, such as radar, LIDAR, GPS, and IMU are grouped to capture the information from the interactions with the vehicle environment. It is worth noting the fact that by using a greater number of sensors of different technologies, a greater degree of robustness of the system is achieved due to the redundancy in the information that can be received, but on the other hand, a greater degree of processing is necessary for data from diverse heterogeneous sources, which increases the computational cost. Liu et al. [70] establish an autonomous lane change (discretionary maneuvering) decision-making model based on benefit, safety, and tolerance functions that analyze not only lane change factors in autonomous vehicles associated with route planning and monitoring, but also in addition to the lane change decision-making process. The benefit function considers the relative speed and distance data between the vehicle and the predecessor vehicles in the same lane and the target lane. The safety function considers a minimum safe distance between the vehicle and the successor vehicle located in the target lane, in addition to the relative distance and speed values between the two. Finally, the tolerance function considers relative distance and speed values between the vehicle and the predecessor vehicle in the same lane, avoiding frequent lane changes if the distance between them is too great. In order to verify the effectiveness of the model in real scenarios, they realized a test verification in the vehicle. The test vehicle used is equipped with Mobileye, millimeter-wave radar, mobile station GPS, AutoBox dSPACE, IMU, and other devices.

In this site are grouped the works that exclusively employ the use of vehicular communications (DSCR) to capture the information from the interactions with the vehicle environment. Although the use of communications manages to fundamentally compensate for the range limitations presented by the use of sensors and cameras, providing greater flexibility in terms of the road safety information that can be exchanged, there are also a series of limitations associated with all technologies of wireless communication, such as packet loss, transmission errors, and communication delay, to name a few. Fallah et al. [71] use the model-based communication scheme in a cooperative FCW system using the example of CAMPLinear and, specifically, the collision detection algorithm proposed in [72] and later refined in [73]. This algorithm uses speed and acceleration information as input, both from the host vehicle and from the remote vehicle, which is the vehicle ahead in its own lane. The model used to estimate the remote vehicle mobility data belongs to the family of follower car models; specifically, the model introduced in [74] is used. The concept of hybrid automation is used [75], which is a well-known method for modeling mixed systems of discrete and continuous states. To evaluate the model-based communications (MBC) approach, two configurations are presented (MBC1 and MBC2), which are compared with traditional communication schemes that directly transmit speed and acceleration data. In both MBC1 and MBC2, each vehicle sends its movement models once at the start of the test and then periodically transmits the updates of the model inputs (speed and acceleration). In the case of MBC2, more sporadic additional messages associated with the change in the movement pattern are also transmitted. It is precisely this second configuration that obtains the most accurate results when tracking the movement of a vehicle.

Huang et al. propose [76] and develop [77] a mechanism based on the real-time estimation of the position-tracking errors of neighboring vehicles, to manage the cooperative information exchange in the vehicle communications environment. They first evaluate decentralized information dissemination policies for tracking-error-dependent multiple dynamic systems and then use collision-error-dependent policy to obtain better tracking performance. Finally, the transmission probability is calculated for each vehicle every 50 milliseconds based on expected tracking errors. Upon receiving information from the channel, each vehicle updates its estimated states of the neighboring vehicles using a firstorder kinematic model, i.e., a constant speed predictor. The main concept of measurements in these algorithms is that the generation of messages from the receiver and the timing of the communication must be determined so that the position-tracking error is reasonably limited. A local copy of the neighboring estimators is executed at each local estimator. The sender compares the output of this simulated estimator with its actual state, thus estimating the position-tracking error, and determines whether remote vehicles need updated messages from the sender. The decision is made by comparing the position-tracking error with a configurable error threshold, generally defined according to the requirements of the cooperative road safety application (RSA).

Mahjoub et al. [78] propose a technology-independent hybrid model selection policy, based on the MBC scheme, for vehicle-to-everything (V2X) communication. The core idea is to implement a hybrid modeling architecture that switches between different modeling subsystems to adapt to the dynamic state of the vehicle. In this particular case, two modeling states are used: one governed by GP, which uses two kernels: one linear and one radial basis function. The other modeling state is defined by the constant velocity kinematic model. A tracking error threshold is used as a selection element when using these models. The results show the effectiveness of the proposed communication architecture both in reducing the required message exchange rate and in increasing the accuracy of remote vehicle tracking. The greater tracking accuracy of the MBC scheme can be attributed to its ability to capture higher-order vehicle dynamics as a result of harsh braking maneuvers and lane change maneuvers.

Mahjoub et al. [79] explore the modeling capabilities of the non-parametric Bayesian inference method: GP, integrated into the MBC design scheme, accurately represents different patterns of driving behavior using only a bank of GP kernels of limited size. To do this, a group of representative trajectories from the SPMD data [80] were selected and the properties of the required kernel bank were explored to be modeled within the GP-MBC scheme. The two fundamental metrics used to evaluate the proposed system are: the length of the transmitted message (related to the size of the kernel bank) and the message transmission rate (related to the persistence of the model). The existence of such a kernel bank allows for transmitting entities to send only the kernel ID instead of the kernel itself, which consequently reduces the length of the prediction, i.e., obtaining a margin of error in the prediction lower than the threshold established by the RSA requirements. The results obtained showed the feasibility of using a group of GP kernels of finite size to predict, with the precision required by the RSA, the future

position of the vehicle through an indirect prediction method, i.e., by predicting the values of the time series of the future speed and direction of the vehicle. An indirect position estimation (speed, direction, or acceleration) achieves superior results compared to direct position estimation.

Vinel et al. [81] design an analytical framework that considers the behaviors of cooperative road safety applications considering the performance of V2V communications. The relationship between the characteristics of V2V communications associated with the probability of packet loss and the packet transmission delay, with the physical mobility characteristics of the vehicle, such as the inter-vehicle safety distance, is analyzed. The case of the cooperative ASV of the emergency electronic brake light defined by the ETSI is analyzed.

Finally, the jobs that use, together with the use of vehicular communications (DSCR), cameras and sensors such as radar, LIDAR, GPS, and IMU are grouped to capture the information from the interaction with the vehicle environment. This favors the complement between communication and sensor technologies, taking the best of both and guaranteeing a more complete performance and greater possibilities for facing the challenges in the area of road safety. Mahjoub et al. [82] design a system for the prediction of the lateral and longitudinal movement of the vehicle. For the prediction of the longitudinal trajectory, nonlinear auto-regressive exogenous models based on neural networks are used. In the case of lateral trajectory prediction, recurrent neural networks (RNN) are used. The system uses two main sources of information: (i) cameras and on-board detection devices such as radars and LIDAR that are assumed as the primary information providers for CAV applications; (ii) V2V communication, which is obtainable using dedicated shortrange communication (DSRC) devices, and is regarded as an important supplementary information source whenever it is accessible. The performance of the system is evaluated not only in ideal communication conditions, but also in the presence of scenarios with up to 40% losses. In this case, a zero-hold estimation method is included to combat packet loss or sensor failure and to reconstruct the time series of vehicle parameters at a predetermined frequency. Its evaluation is simulated using real-communication scenarios with data extracted from safety pilot model deployment (SPMD) [80]. Du et al. [83] develop a network architecture called vehicular fog computing to implement the cooperative data census of multiple adjacent vehicles circulating in the form of a platoon. Based on this architecture, a greedy algorithm is used to maximize the census coverage (associated with the area ratio and total ratio parameters) and minimize the overlap coverage (associated with the efficiency parameter), enhancing the parallel calculation through of the distributed management of the computational resources of the platoon members. A SVM algorithm is used to merge the census data of multiple vehicles and obtain precise information on the status of the vehicles. Through an occupancy grid filtering (OGF) of the on-board sensors (LIDAR, cameras), the environment is mapped as occupancy states. These OGF maps are integrated into the head vehicle, and by means of SVM it is classified when a grid is occupied by a vehicle. To train the SVM classifier, the GPS position data extracted from the NGSIM database is used to locate the vehicles on the OGF map. Finally, the result of the merger of the census data of multiple vehicles (location of the vehicles on the OGF map) feeds the algorithm of a light GRU neural network to predict the discretionary maneuvers of lane changes to classify them into lane-keeping maneuvers or lane change maneuvers.

Moradi-Pari et al. [84] use the model-based communications scheme to design a small-scale and large-scale modeling strategy for the dynamics of vehicular movement. The representation model of the system to describe the behavior of the vehicle is based on the representation of stochastic hybrid systems, where: (i) small-scale evolution represents actions of braking and acceleration, represented by exogenous auto-regressive models and (ii) large-scale evolution, which includes lane change maneuvers and free circulation flows, are represented by coupling these models within a Markov chain. At each model calculation time, all currently available states of the latest version of the model must be explored, and the best-fit parameter values must be found for each of them according to the

new observation element. If at least one of these states satisfies the error threshold specified by the application using its new parameter values, the current model is assumed to be fully descriptive for the entire observation sequence received. However, if the minimum error reached given by the current model exceeds the required threshold, it is necessary to introduce a new state to represent the new observation segment and describe the last maneuver of the driver. To evaluate the performance of the proposed models and adaptive cruise control methods, various scenarios were simulated with different realistic data sets, including data from SPMD [80] and driving cycles for Environmental Protection Agency testing standards [85].

We want to emphasize that cooperative driving can significantly contribute to the development of C-ADAS, since this is the most important result of detection, communication, and automation technologies, and, in turn, significantly influences the behavior of drivers. Z. Wang et al. [37] provide a review of the literature associated with cooperative multiple CAV longitudinal motion control systems, with an emphasis on the architecture of several cooperative CAV systems. An in-depth discussion of the control aspects of CAV systems is carried out, highlighting the main challenges generated by the existence of different information flow topologies, which are mainly focused on string stability, communication issues, and dynamics heterogeneity. Zhou et al. [86] present a literature review of learning-based longitudinal motion planning models for autonomous vehicles, focused on the impact of these models on traffic congestion. They surveyed the non-imitation learning method and imitation learning method, and the emerging technologies used by the principal automakers for implementing cooperative driving are described.

Multiple research works have investigated in the context of cooperative driving the design of control systems that favor the management of traffic and/or the crossing of intersections. Zu et al. [87] propose a cooperative method for connected automated vehicles that controls the timing of the traffic lights and manages the optimal speed at which the vehicles should circulate. The optimization of the traffic light times and the calculation of the arrival times of the vehicles at the intersection allows for the minimization of the total travel time for all the vehicles, as well as the fuel consumption of the individual vehicles. Zheng et al. [88] establish analytical results on the degree of stability, controllability, and accessibility of a mixed-traffic system composed of autonomous vehicles and human-driven vehicles. The proposed system allows the flow of traffic to circulate at a higher speed and shows that the autonomous vehicles, along with cooperative driving, can save time and energy, smoothing traffic flow and reducing traffic undulations. Wang et al. [89] propose a cooperative platoon system for CAV, based on a predictive control model with real-time operation capability, to efficiently manage the vehicle tracking behaviors of all CAVs in a platoon. The constant time advance method is used to adjust the balance gap between successive vehicles. Zhou et al. [90] introduce a smooth-switching control-based cooperative adaptive cruise control scheme with information flow topology optimization to improve riding comfort while maintaining string stability. A Kalmann filter-based predictor is used to estimate the state of the preceding vehicle, suppressing the noise in the measurement and estimating the acceleration of the vehicle in the event of communication failures. Zhou et al. [91] propose a hybrid cooperative intersection control framework to manage the entrance and exit of a group of vehicles to an intersection. A virtual platoon is defined to group these vehicles according to their proximity to the entrance of the conflict zone of the intersection. The location assignment of the vehicles within the virtual platoon differs from their real relative locations. This virtual platoon is obtained by linearly projecting the distances at which the vehicles are from the center of the intersection, then, platoon control rules are applied to manage the movement of vehicles approaching the intersection.

The management of the flow of information exchanged in a V2V communication environment has been addressed in [92]. Wang et al. develop a mathematical modeling based on queues to manage the transmission of information of multiple classes with different levels of delay according to the cooperative road safety applications. In [93], these authors also addressed traffic management in mixed environments with the presence of human-driven vehicles and autonomous and connected vehicles, analyzing the differences in the principles of route choice and the traffic patterns followed by human-driven vehicles and connected and autonomous vehicles, and they also considered the use of preferential circulation lanes with free access to connected and autonomous vehicles in [94].

Table 3 summarizes the main works consulted that address elements of the vehicle– environment interaction. The directionality in which this interaction is approached is analyzed, as well as the way in which it is implemented.

Table 3. Works in which vehicle–environment interaction is addressed. The "ideal" behavior means that the authors do not consider the losses of the communication channel. The works that are not grouped within "ideal" behavior or "non-ideal" behavior, are those in which the use of communications is not considered for the design of ADAS.

Articles	es Directionality		Implementation of the Interaction				Communications	
	V -> E	E -> V	Data Source	Data Type	Environment	"Non-Ideal" Behavior	"Ideal" Behavior	
[82]		\checkmark	Cameras, radar, LIDAR, and DSRC	Real	Real	\checkmark		
[66]		\checkmark	Radar	Simulated	Simulation			
[55,57-62]		\checkmark	Cameras	Real	Real			
[63]		\checkmark	Cameras	Real	Simulation			
[68]		\checkmark	LIDAR	Real	Real			
[69]		\checkmark	Radar	Real	Real			
[70]		\checkmark	Cameras, radar, GPS, and IMU	Real	Real			
[83]	\checkmark	\checkmark	Cameras, LIDAR, and DSRC	Real	Real and simulation		\checkmark	
[71]	\checkmark	\checkmark	DSRC	Real and simulated	Simulation	\checkmark		
[65]	\checkmark		Cameras	Real	Real			
[84]	\checkmark	\checkmark	Cameras, radar, LIDAR, and DSRC	Real and simulated	Simulation		\checkmark	
[76,77,81]	\checkmark	\checkmark	DSRC	Simulated	Simulation	\checkmark		
[78]	\checkmark	\checkmark	DSRC	Real and simulated	Real	\checkmark		
[79]	\checkmark		DSRC	Real	Real	\checkmark		
[37]	\checkmark	\checkmark	DSRC	Real and simulated	Real and simulation	\checkmark		
[86]		\checkmark	Cameras, radar, and LIDAR	Real and simulated	Real and simulation			
[87]	\checkmark	\checkmark	DSRC	Simulated	Simulation	\checkmark	\checkmark	
[88]	\checkmark	\checkmark	DSRC	Simulated	Simulation		√	
[89]	\checkmark	\checkmark	Cameras, radar, LIDAR, and DSRC	Real and simulated	Simulation		\checkmark	
[90]	\checkmark	\checkmark	Cameras, radar, LIDAR, and DSRC	Real and simulated	Simulation	\checkmark		
[91]	\checkmark	\checkmark	DSRC	Simulated	Simulation	\checkmark		
[92]	\checkmark	\checkmark	DSRC	Simulated	Simulation	\checkmark		
[94]	\checkmark	\checkmark	DSRC	Simulated	Simulation	\checkmark		
[93]	\checkmark	\checkmark	DSRC	Simulated	Simulation	\checkmark		

4.2. Current Challenges

The addressed challenges in this section are focused on the sensors and communications technologies. The main challenges with regard to sensors technologies are:

(i) The sensor occlusion with respect to the line of sight of the objects and other road actors. The good performance of these technologies is affected under various climatic and environmental conditions, such as roads with markings covered by snow, heavy rain, or dense fog. Objects, people, and animals located in the vicinity of the vehicle, or

obstructing each other, represent serious security problems for detection by these devices. This phenomenon is not so serious when the objects are located at greater distances, where the processing algorithms can help the sensing devices to improve detection tasks. These drawbacks can be minimized by sensor redundancy as in [95], where a 360-degree vision system is used for parking assistance.

(ii) The high computational resource consumption of image processing algorithms present in camera-based sensors. Detection at distances greater than 200 meters requires the use of ultra-high-resolution cameras for the sensing of small details in the target image. Therefore, powerful image processing algorithms are required to analyze the high volume of image data and extract useful information from the noise associated with it [96], which continues to represent a limitation to the adequate processing in real time that road safety demands.

(iii) The high cost of specific hardware technologies. Some well-established technologies in the market, such as vision cameras or radars, have managed to establish themselves in large-scale production, lowering their production costs. This situation, however, differs from other more specific technologies for the automotive industry, such as LIDAR devices, whose standard incorporation in vehicles considerably increases their price depending on various factors, such as the type of use for the one that the vehicle is destined for, or their sensing capabilities in 2D or 3D.

The main challenges with regard to communications technologies are:

(i) The scarce deployment of network infrastructure in the road environment. The massive implementation of V2I technologies is an expensive and time-consuming task. Various costs must be assumed depending on the location environment, such as the installation of nodes, bandwidth, and energy support, and the subsequent maintenance of the installed equipment. An important aspect is the traffic capacity of these networks, given that cooperative road safety applications require a high degree of penetration of vehicles with installed connection capabilities, e.g., in the case of the cooperative collision warning application, a density greater than 60% of the total connected vehicles is desirable [97].

(ii) The lack of robustness of current communications networks to operate in a vehicular environment, evidenced by the loss of transmitted packets and other problems, such as communication channel congestion, transmission delay, and fading and shadowing in signal propagation. The reliability and accuracy of the exchanged data is vital to ensure the proper functioning of cooperative applications, as inaccurate data can result in bad judgment when making a decision related to road safety. These networks must guarantee robustness against sending duplicate messages or false positives in the issuance of alerts. Furthermore, even when the data exchanged is accurate, the freshness of the information is required, due to the importance of the temporal component of road safety data. It is not enough to detect a risk situation on the road and communicate it accurately, it is also required that this information arrives in time for the appropriate reaction to said danger and thus guarantee the good global operation of cooperative road safety applications. In this sense, emerging technologies such as 5G can provide greater capacity and reduced latency when exchanging this sensitive information [98]. With the progressive increase in the volume of data, the need arises to have tools and algorithms capable of efficiently processing them. Raw data handling is becoming less feasible and more expensive. Dimensionality reduction techniques are required to identify patterns in data and achieve more scalable developments [99].

(iii) The security and privacy of the information exchanged represent an important challenge to consider when designing C-ADAS. As in any communication network where private information is exchanged, the issue of data security and integrity is vital, but in these networks in particular it is even more important, since people's lives are at stake, as well as overall road safety. Hacking activity in networks without adequate protection can cause attackers to take control of a vehicle's security systems, causing traffic diversions or the activation of the emergency braking system, the stealing of personal data from users, and in the worst case scenario, the creation of conditions for the occurrence of traffic accidents. These challenges are part of the OEM and after-sales connectivity systems [100], which is

especially sensitive in automated vehicles with electronic control of their actuators. Another current problem is that network security approaches can, on occasion, compromise vehicle security due to the overload of exchanged security information, generating excessive delays in communications, which is associated with authentication mechanisms, validation, and security certificates.

The shortcomings that still persist, and the study and achievements related to the V–E interaction, are the inability to operate bidirectionally, the lack of holistic integration between the detection, communication, and processing technologies, as well as the joint operation with the driver subsystem, which allows for the exploitation of these redundant pathways to obtain information from the environment. In addition to this, there are also the challenges associated with the functioning and operation of these sensing (e.g., resolution and scope) and communication (e.g., reliability and delay) technologies, which undoubtedly limit the effectiveness of the current implementation of the proposed C-ADAS.

5. Driver-Vehicle (D-V) Interaction

The analysis of bidirectionality when studying the interactions between driver and vehicle (D–V) is relevant for proactive decision making in a vehicular context. Let us consider an example where the driver is developing aggressive driving on the road—the C-ADAS detects this behavior (D -> V interaction). This characterization of the driver will allow the C-ADAS to estimate possible risk situations when this vehicle approaches other aggressive or extremely conservative drivers. Given such knowledge, convenient and personalized messages could be sent to moderate the conduct of those involved, which is a more complex form of interaction (D -> E -> V). This redundancy alternative path is relevant to consider, given that HMI devices have limitations and challenges for optimal performance in cooperative environments and sometimes limited customization features. This degree of redundancy can only be achieved if the design of the C-ADAS is approached from a holistic and systemic vision, with bidirectional interactions among subsystems.

The D–V interaction describes the main elements related to the HMI module. This area is in charge of presenting the driver with the information about the environment acquired by the sensors and communication technologies and the information of the vehicle itself, obtained from the internal communication pathways, such as the CAN bus. Similarly, it is also necessary to sense the driver's reaction to the operation of the assistance system. At the same time it is in charge of materializing, through the vehicle operation, the driver's actions on the environment that surrounds him. As an "assistance" system, its first function should be to learn about the particular characteristics of the driver, understand their actions, and to be able to model their behavior (instantaneous and historical) to identify the "particular assistance needs" that the driver requires. It must also be able to communicate that knowledge acquired to the rest of the actors on the road, as part of the concept of cooperative knowledge. Note that, the warnings and vehicle take-over should be limited to situations evaluated with a high degree of certainty and personalization. The redundant information in terms of sources, modules, and communication paths, as well as the prior and continuous characterization of drivers and their driving styles, prevent the unnecessary modification of driving styles and regulates those aggressive behaviors that represent a danger.

5.1. Actual State

The discussion of this section begins by grouping the works that use the devices that allow the driver to operate on the vehicle while driving to capture the information of the driver–vehicle interaction, obtained through the internal communication buses of the vehicle, for example, the CAN bus. Although this information is present in most modern vehicles, it is not currently sufficiently exploited by the automotive industry to establish a personalized interaction between vehicle and driver. Wang et al. [101] propose a method to predict the driver's braking intention in car-following scenarios from a perception–decision–action perspective according to their driving history, considering the following variables:

vehicle speed, distance between the host vehicle and vehicle preceding, and the relative speed between them and TTC. The system combines a Gaussian mixture model (GMM) with a hidden Markov model (HMM) to infer the driver's braking action given the state of the driving situation, using data from the CAN bus, cameras, and radars installed in the vehicle. Dang et al. [102] design an adaptive cruise control system with a lane change assistant. First, the risk associated with the lane change is analyzed by calculating the minimum safe space between the host vehicle and the surrounding vehicles. To calculate this risk, a driving style factor is introduced, which modifies the calculation of the minimum safety distances between the vehicle performing the maneuver and neighboring vehicles. The value of this factor is set by the driver. Values greater than one indicate conservative conductors, lower values indicate aggressive conductors. Finally, a coordinated control algorithm is developed using a predictive model control theory that limits the longitudinal acceleration of the vehicle to guarantee a better performance in terms of comfort in movement.

Zhu et al. [103] propose a personalized driver assistance system that includes driver profile identification for lane change assistance. Initially, the data obtained through driving simulators are analyzed and statistically processed to select the most relevant variables. With these results, the fuzzy c-means clustering algorithm is used to extract different conduction profiles. Three clusters are generated, which are associated with aggressive, normal, and conservative profiles. A neural network classifier, optimized by a particle swarm algorithm, is used to detect these conduction profiles. According to the profile of the driver identified by the system, preset values are determined for the execution time of the lane change maneuver and the minimum safety distance margin required with the preceding vehicle in the lane to which the change is made to avoid a forward collision. These personalized values are included in the analytical model to calculate the risk of collision associated with lane change and the subsequent behavior of the following vehicle in the destination lane. Su et al. [104] design a forward collision system that employs a method to recognize driver intent and driving behavior, based on the GMM. The proposed system has the advantages of adapting the model and the ability to generate probability densities in arbitrary shapes. In addition, it allows real-time implementation and has high precision. A precise recognition model of the driver's driving behavior is first established and verified in a real-time driving simulator with 36 drivers as samples. Then, an FCW algorithm is designed with a braking execution strategy and an alarm classification based on the results of the driver's driving behavior recognition. Drivers are classified as conservative, normal, and reckless. Mantouka et al. [105] identify driving styles based on unsupervised classification techniques, using acceleration and speed data collected through the use of smartphones. Initially, the styles are grouped into aggressive and non-aggressive behaviors to subsequently analyze additional unsafe behaviors associated with distraction and risk taking. Once the driving profiles have been detected, the driver's average behavior and its persistence or volatility in different situations are analyzed.

Works that use the vehicle driving devices to capture the information of the drivervehicle interaction have been grouped here, including the use of devices that emit visual, sound, or vibro-tactile alerts to the driver and analyze their reaction to these alerts. Although the use of devices that alert the driver is increasingly being addressed by the automotive industry in modern vehicles, the personalized interaction between vehicle and driver, which also includes the issuance of personalized warnings according to the driver's behavior, continues to be a challenge in this area. Yang et al. [106] propose a collision warning system based on V2V communication. The algorithm initially detects the intention of the driver of the preceding vehicle, which is transmitted, together with other movement parameters of said vehicle, to the following vehicle through the V2V communication module. Finally, the safety application that runs in the following vehicle estimates the risk of potential collision with the information received through V2V and with the movement parameters of the vehicle itself, to alert the driver, outperforming systems based on TTC thresholds. Bavendiek et al. [107] present a human–machine interface design method based on the concept of metaphors to analyze and improve the vehicle–driver interaction through the design of HMI interfaces. This methodology, whose best known example is the design of computing machines based on the desktop interface, enables a friendlier relationship between man and machine, achieving improvements in the design of the HMI interface installed in the vehicle. The study focuses on describing a procedure to identify metaphors in the HMI environment of the automobile. Subsequently, the development of new HMI concepts based on the identified metaphors is proposed.

Iranmanesh et al. [108] design a FCW system that considers the driver's historical braking profile in the face of previous alert events to establish a reaction threshold. The time of advance (time headway) for a vehicle is understood as the time it would take for the vehicle to travel circulating at its current speed and the distance between its front part and the front part of the vehicle that precedes it. The main metric is the alert activation, which is defined as a warning-triggering threshold, considered as the normal level of risk tolerance, to determine the need to issue an alert to the driver in the event of a possible dangerous situation. The caution-deceleration threshold is also defined to avoid false alerts and discard situations outside a dangerous situation, such as stops or turns in the presence of traffic lights. The system continuously monitors driver distraction through data obtained from the CAN bus associated with the state of the acceleration pedal, speed, acceleration, and turning angle, among others. The detection of driver distraction is performed by SVM classifiers and a multi-layered perceptron neural network. Sun et al. [109] propose a lane change assistance system based on the identification of the individual driving profile, determining an optimal alert threshold that varies as the characteristics of the individual profile change. The authors use the signal detection theory (SDT) to develop a method to determine the characteristics of the driver in a lane change maneuver. The target signal is defined as the operation to complete the maneuver by the driver and the noise signal is the operation to abandon the maneuver, which are associated with warning and non-warning criteria for the lane change warning system. The warning threshold is adjusted in real time, according to the particular characteristics of each driver during the maneuver. According to the authors' definition, those who complete the lane change maneuver even in the presence of the safety system warning signal are grouped as aggressive drivers, and those who abandon the lane change execution even in the absence of the system alert signal are grouped as conservative drivers. Based on the analysis of the existing warning criteria, variables such as the TTC and the relative distance between the subject vehicle and the rear vehicle in the destination lane were used as warning indicators, while the initial warning threshold according to the difference of the speed of the subject vehicle was selected for the adaptive algorithm.

Choi et al. [110] propose a personalized design of the next generation HMI interfaces, where the driver can personalize the way in which he/she interacts with the vehicle and, in turn, it responds in a personalized way, identifying the characteristics and the state of the driver in a given situation. The proposed system consists of elements such as sensors embedded in the car, an adaptive inference engine that analyzes the driver–vehicle interaction, and an advanced digital platform in the vehicle cabin, which accesses the data obtained from this interaction. Dargahi Nobari et al. [111] propose a control scheme with feedback that considers the state of the driver as an input element for the system that analyzes the driver–vehicle interaction. Sensors (e.g., eye-tracker, physiological sensors) are used to detect the state of the driver. This result is compared with previously established situations to then design a policy that regulates the generation of stimuli tending to reduce the degree of criticality of the traffic situation. Table 4 summarizes the main works consulted that address elements of the driver–vehicle interaction. The directionality in which this interaction is approached is analyzed, as well as the way in which it is implemented.

Table 4. Works in which driver–vehicle interaction is addressed. The type of personalization described in the consulted works is analyzed from two approaches: (i) the action that the driver exerts on the driving elements of the vehicle (described as an action on the vehicle) and (ii) the reaction that the driver adopts before the notices and alerts sent by the system (described as reaction to ADAS).

Articles	Directionality		Implementation of the Interaction			Personalization Types	
	V -> D	D -> V	Data Source	Data Type	Environment	Action on the Vehicle	Reaction to ADAS
[101]		\checkmark	CAN bus	Real	Real	\checkmark	
[102]		\checkmark	CAN bus virtual	Real	Simulation	\checkmark	
[103,104]		\checkmark	CAN bus virtual	Simulated	Simulation	\checkmark	
[105]		\checkmark	Smartphone sensors	Real	Real	\checkmark	
[106,107]	\checkmark	\checkmark	CAN bus and warning indicator	Real and simulated	Real and simulation	\checkmark	\checkmark
[108–110]	\checkmark	\checkmark	CAN bus and warning indicator	Real	Real	\checkmark	\checkmark
[111]	\checkmark	\checkmark	Eye-tracker and warning indicator	Real	Simulation	\checkmark	\checkmark

5.2. Current Challenges

The current challenges of this section are focused mainly on the personalization of the ADAS and the motion modeling.

(i) The personalization of the ADAS, which should consider the driving profile, preferences, and peculiarities of the driver. In the design of HMI devices, not only the driver's action on elements of the vehicle that determine its state of movement should be considered, but also feedback elements through visual or sound information that alert the driver about the consequences of their actions and that of the other actors in the environment regarding road safety. There is a common assumption in the personalization of ADAS systems that the driver is more comfortably adjusted to systems that implement a driving style similar to their own, but in practice, determining that optimal driving style for each individual driver is a very challenging task. In current systems, the process of interaction between driver and vehicle through the HMI limits the driver's ability to correct and shape the system to establish a driving style that provides greater comfort. This interactive exchange phase between vehicle and driver requires further development. Another aspect to deal with in greater depth is the fact that the personalization process must be conceived as a continuous process. It is not enough to limit the customization process to obtaining and establishing a personalized system, since the drivers, influenced by various internal and external factors present on the road, can modify their preferences and driving styles in certain situations. For this reason, this phase of interactive exchange between driver and vehicle must last over time as a continuous process, improving its usability characteristics and finally, its benefit for road safety.

(ii) Modeling the longitudinal and lateral motion is a challenging task, which includes predicting the driver's intention in response to the dynamics of the surrounding environment. The accurate sensing of the surrounding environment and the prediction of the intent of neighboring vehicles represent the biggest challenges for modeling the lateral and longitudinal motion of a vehicle by considering human driving preferences in the process. When the distances between vehicles, pedestrians, and objects are small, the degree of precision of on-board sensors must be very high. Similarly, if we take into account that the movement of vehicles on the road is a complex scenario of interaction between several actors, it is crucial to have tools that allow us to infer the intention and future behavior of neighboring vehicles. The failure of any of these two elements can lead to a tragic result, the dynamic instability of the vehicle, and even a traffic accident. When dealing with a high risk of collision, the development of conservative algorithms is chosen, even if this sacrifices aspects of the system such as efficiency, comfort, and acceptance of drivers and passengers. Incorporating V2V and V2I communications into the system can overcome these limitations and inefficiencies of non-cooperative systems [112].

(iii) Vehicles with SAE level three automation allow the driver to freely participate in tasks not directly associated with driving, but the driver is required to be able to disconnect from these tasks to regain manual control whenever required by the system. This requirement, related to the term take-over, represents a challenge to consider in the study of D–V interaction. This capability demands that the driver carry out this process of returning to driving tasks, ensuring a smooth, and at the same time, safe transition towards taking control of the vehicle. This challenge requires a novel design when addressing the D–V interaction (between the driver and the automated system), which considers the driver's ability to take control in real time [113]. Situational awareness may be diminished in highly automated driving environments compared to manual driving environments if drivers are engaged in non-driving tasks [114]. In [115], the authors present a novel predictive haptic take-over controller to further explore the safe and smooth interaction mechanism during the take-over of autonomous vehicles.

The works described do not show the required bidirectionality between driver and vehicle, nor a holistic integration between subsystems, which is essential for information redundancy and system reliability. However, the more specific challenges in D–V interaction are associated with the degree of customization of these HMI devices for the timely display of alerts to the driver, the high-precision modeling and prediction of the driver's intentions and the execution of the maneuver, and the guarantee of safe take-over transitions. Note that these interaction challenges not only limit the implementation of the proposed C-ADAS but also the progress in autonomous driving.

6. Driver-Environment (D-E) Interaction

The analysis of bidirectionality in the study of the interaction between driver and environment D–E is relevant for the C-ADAS to be aware of how attentive drivers are to the environment and how they react to eventual risk situations. Let us analyze, for example, the information provided from the environment directed towards the driver, which can be a stop warning signal. This information can be received: (i) directly by the driver $(E \rightarrow D)$, or (ii) indirectly, via a stop notification issued from an RSU and notified by the vehicle to the driver ($E \rightarrow V \rightarrow D$). Regardless of the way in which the driver perceives the stop order, the C-ADAS must guarantee the vehicle's stop. In this sense, it is vital to determine as soon as possible if the driver is aware of their surroundings and will carry out the braking maneuver effectively. For this purpose, the C-ADAS must continuously monitor the driver behavior through sensors, cameras, and wearable devices on board the vehicle (D -> V). In this particular case, if the driver does not react adequately, then the C-ADAS itself can issue an alert to stop, contributing as a redundancy path that guarantees that the information that the driver should receive is received satisfactorily ($D \rightarrow E, D \rightarrow E$), but if the risk situation cannot be fully mitigated by warning or vehicle take-over, then a general stop warning should be issued to alert other drivers and vehicles of the situation (D -> V -> E). It is important to take this redundancy route into consideration by the C-ADAS, given that there may be innumerable occasions in which the driver fails to perceive relevant information in terms of road safety, either due to distraction or due to impairments in their driving operational capacity. This degree of redundancy can only be achieved if the design of the C-ADAS is approached from a holistic and systemic perspective, where the study of bidirectionality in interactions is considered. The works of the state of the art, which are described in this section, fundamentally contribute to the indirect ways of obtaining this information and, in this sense, they are relevant in terms of the use of sensors, cameras, and wearable devices to monitor the status and behavior of the driver.

The importance of the natural perception's module lies in the fact that the drivers form a personal reality about their surrounding environment, and make their own assessment of risk situations and the interaction between the different actors on the road, and based on this, the module determines their primary behavior before interacting with the ADAS. This module of natural perception has an amazing capacity to process large volumes of very diverse data, and this versatility of processing in humans still represents a challenge for the computer systems that exist in the present. However, it also has real limitations regarding the level of precision when estimating the kinematic variables associated with the movement of vehicles, something that is vitally useful in potential-risk situations on the road. This can be seen reflected in the deficient calculation of the adequate distances to carry out lane change or overtaking maneuvers, in maintaining an adequate safety distance with the vehicle in front, and detecting the presence of vehicles approaching from behind in adjacent lanes, among other common situations that arise daily. As an active element within the dynamics of the road environment, the drivers modify the state of said environment through their actions and reactions. The use of various signals by the drivers represents how they interact with the surrounding environment. Among these, we can mention: (i) visual signals through the vehicle's lighting devices, such as turn signals, position indicator lights, or the stop lights; (ii) sound signals through the use of the horn; and (iii) body signals through the gestures or so-called "hand signals" of the drivers. The technological development associated with the infotainment area, together with the challenges imposed by the increase in the complexity of the road infrastructure and the increase in traffic congestion, have shown a growing use by drivers of navigation applications through the use of maps for route planning, which to some extent conditions their future behavior and mobility.

6.1. Actual State

The discussion of this section begins by grouping the works that use cameras to capture the information of the driver-environment interaction, a method that is currently widely used in the automotive industry to analyze the behavior of the driver inside the vehicle. Qiao et al. [116] propose a fatigue detection system using images of the driver's face, eyes, and mouth, obtained by a smartphone camera. Signals of fatigue such as the blinking of the eyes are detected by means of a Haar qualifier [117], while sudden movements of the head are captured by calculating the variance of the centroid of the face. Yawning is detected by measuring changes in the geometry of the mouth through the Canny active contour method [118]. Mandal et al. [119] propose a fatigue detection system for bus drivers, using a percentage of eyelid closure (PERCLOS) method to determine the level of eye opening. Initially the system locates the position of the driver's head in the incoming image to detect the location and orientation of the eyes. Yuen et al. [120] propose a system to monitor the driver's activity during driving, analyzing information related to facial reference points, which is used for the detection of the face and the position of the head. Its performance is analyzed under various lighting conditions and degrees of occlusion of the driver's face, which enables the system to be able to detect when there are occluded parts of the face and consequently achieve better estimation results in this situation.

Next, the works that use smartphones to capture the information of the driverenvironment interaction are grouped together, a method that allows, in addition to the use of the smartphone camera, the use of the various sensors embedded in it, which allow for an obtaining of various physical variables that favor a more complete analysis of the driver's actions while driving. One of the principal causes of vehicle accidents is distraction during the driving process. Eraqi et al. [121] present a vision-based system that uses RGB images obtained from the rear camera of a smartphone to recognize distracted driving postures, composed of a face detector, a hand detector, and a skin segmenter. The proposal is implemented using convolutional neural networks (CNN), obtaining results of the order of 90% accuracy; however, its performance overhead is higher in a real-time configuration. Janveja et al. [122] present a smartphone-based system for the detection of driver distraction by analyzing gaze tracking through the left and right rear-view mirrors and for fatigue detection by monitoring yawning and eye-blinking frequency. The system is designed to operate in low-light conditions using two configurations: in the first, a near-infrared (NIR) LED coupled to a smartphone is used; in the second, a generative adversarial network is used to synthesize a thermal image obtained from the RGB camera of a smartphone. The results show a better behavior of the system when NIR images are used. Kapoor et al. [123]

designed a smartphone-based driver distraction detection system capable of operating in real time, which alerts the driver with a beep once distracted behavior is detected. The 10 classes of distracted behavior are drawn from the State Farm Distracted Driving Database [124], which is used for fine-tuning the four pre-trained CNN models, namely MobileNetV1, InceptionV3, VGG-16, and MobileNetV2. Xie et al. [125] present a system for the detection of driver distraction based on the use of data from GPS and the IMU sensors of a smartphone during the performance of turning maneuvers, lane change, lane maintenance, stops, and near stops. According to the results published by the authors, the best performance of the distraction detector, in terms of the F1 score metric, is obtained for lateral maneuvers (turn and lane change). This is assumed to be because the data used are more sensitive to this type of movement. This metric is defined by the authors themselves, using sliding windows to extract the temporal characteristics of the data obtained from the sensors and weighting the classification result of each sliding window according to the number of normal driving or distracted driving labels.

Next, we describe a work that uses the devices and internal sensors of the vehicle, which are grouped to capture the information of the driver–environment interaction, through physical variables that they describe as the result of the driver–environment interaction directly in the vehicle. Hu et al. [126] present a system to detect abnormal driving behaviors, such as recklessness, fatigue/drunkenness, and use of a smartphone. Unlike other works, where the driver's activity is monitored by cameras that capture the image and movements of the driver while driving, in this study the authors analyze vehicle movement patterns, such as sudden acceleration and braking with a delayed response to traffic conditions, to detect abnormal driving behavior. To quantitatively assess these behaviors, a driver abnormality index is proposed. Qi et al. [127] present a passenger and driver activity detection system by means of acoustic recording devices for the recording and inference of activity inside the vehicle and by means of IMU and GPS sensors, including OBD-II [128] data for human activity detection and for the detection of vehicle movement patterns, such as braking, lane changes, and turns.

The works that use wearable devices and biomedic sensors have been grouped to capture the information of the driver-environment interaction, which allows a deeper analysis from the biological and physical point of view of the behavior of the driver during driving. The use of these devices, however, presents as a challenge the design of less invasive systems for the driver, with the aim of not distracting them or making them uncomfortable during the driving process, together with the necessary medical restrictions so as not to compromise the driver's health. Rohit et al. [129] exploit the use of wearable EEG sensors for the real-time detection of driver drowsiness. An SVM classifier is used to detect drowsy states, by means of a spectral analysis of the EEG signals obtained from the drivers. Additionally, the blink duration parameters are extracted and analyzed, which were less favorable than the spectral analysis for the detection of drowsiness. Li et al. [130] develop a system for the early detection of driver drowsiness using not only the signal obtained from a wearable EEG sensor, but also by incorporating a gyroscope coupled to the driver's head to analyze the movement of the driver's head. Different from previous works, this study analyzes the feedback of the system when stimulating the driver by means of transcranial direct current to improve their state of alertness in real time while driving. In the same way, visual and vibro-tactile alerts are presented to the driver to combat the different levels of drowsiness detected by the system. Guo et al. [131] present a study to analyze the transition process of the driver's intention, caused by the driver's emotions. Various visual, olfactory, and auditory stimuli are used to generate emotions in the driver before driving tests and maintain them during the driving process. The results show a high accuracy and reliability to estimate the driver's intention through the evolution of their emotions, and this system can be used to design personalized driving alerts in the human-machine interfaces of modern vehicles.

Finally, works have been grouped in which the driver plans their future route through the selection of points of origin and destination, which may undergo modifications according to the dynamic characteristics of the environment. Through the ADAS assisted by navigation maps, the driver is an active agent that modifies the environment, and through route planning, the state of the road environment is modeled. An assistance opportunity could be designed based on specific sections present in the routes defined by the driver, considering traffic between roads with different vehicle-flow capacities, entering roundabouts, overpasses, joining or exiting motorways, lane changes, and turns at intersections, among others. All these maneuvers involve risk, and through a cooperative knowledge of the pre-established routes, this risk can be minimized at these critical points for road safety. In these cases, an explicit notification prior to carrying out maneuvers at these points could mean the difference between the occurrence or not of a road accident. Xu et al. [132] present a study focused on improving intelligent transport management to combat the problem of predicting road congestion levels in real time. To do this, they develop an analysis method implemented in big data and cloud-computing platforms, which enriches the traditional method based only on the use of historical driving data, while also incorporating the users' travel plans contained in the vehicle navigation information, which is associated with route planning. The system visualizes the data from this analysis by means of heat maps and sends personalized notifications to drivers according to the particular situation of their road environment.

Withanage et al. [133] develop a personal navigation system that simplifies the user's interpretation of the translation of voice commands and the visualization of the routes in the navigator. During this process, the system initially translates the audio files into text using automatic speech recognition (ASR), and then uses natural language processing (NLP) techniques to retrieve previously undetected navigation information, finally displaying the generation of trajectories on the map using the development interface provided by Google Maps. Keerthana et al. [134] designed a navigation assistant based on voice instructions as a human-machine interface to guide the user to the requested destination through text-to-speech techniques to show the source and destination addresses on the map, allowing the planning algorithms to obtain route information by recognizing the user's voice instructions. The addition of voice recognition techniques represents an improvement to the navigation tools for an individual client who needs to navigate in an obscure landscape. Zhou et al. [135] propose a system for planning tourist routes by correlating data associated with tourist places with the precise data of personal interest of the individual. To do this, it studies the behavior and personal needs of tourists and, based on this, it proposes tourist places with views and characteristics that are related to the interests previously analyzed. This method allows to obtain a more personalized route planning and its viability is testified through the design and execution of experiments with real data. Rathnayake et al. [136] present an interactive system for planning and evaluating travel routes, which analyzes the distances and weather conditions of the moment to carry out an evaluation of the trip previously established by the user. As a result of this analysis, recommendations and improvements to the initial travel plan are established, guaranteeing the optimization of the journey in terms of the distance and coverage of the different destinations selected a priori by the user.

Here, we consider driving behavior by capturing and detecting from on-board sensors of the bodily and physiological reactions that the driver manifests regarding stimuli from the environment. The knowledge of the D–E interaction can be correlated with the knowledge of the D–V interaction to be able to conclude about the ability or not of the driver to respond to certain events. Through V2V communication, the C-ADAS knows that the ego vehicle is approaching the vehicle in front of it (E–V interaction), simultaneously, through on-board cameras, it perceives drowsiness in the driver (D–E interaction) and, in turn, low pressure on the brake (D–V interaction). Consequently, the intervention of the C-ADAS in driving is required to avoid the accident, given the inability of the driver to react appropriately. Table 5 summarizes the main works consulted that address elements of the driver–environment interaction. The directionality in which this interaction is approached is analyzed, as well as the way in which it is implemented.

Table 5. Works in which driver–environment interaction is addressed. In this work, we decided to analyze two of the characteristics most addressed in the literature consulted on detection of the driver state: (i) monitoring of driver fatigue or drowsiness (fatigue monitoring) and (ii) driver distraction. Furthermore, we analyze the works in which the driver has an active role in the modification of the environment's state through the planning of their future route.

Articles	s Directionality		Implementation of the Interaction			Driver Status	
	E -> D	D -> E	Data Source	Data Type	Environment	Detection of the Driver State	Trajectory Planning
[116,119]	\checkmark		Camera	Real	Real	Fatigue monitoring	
[129]	\checkmark		EEG and wearable sensors	Real	Simulation	Fatigue monitoring	
[130]	\checkmark		Bluetooth sensor and smart watch	Real	Real	Fatigue monitoring	
[126]	\checkmark		CAN bus	Simulated	Simulation	Fatigue monitoring, driver distraction	
[121,123]	\checkmark		Smartphone camera	Real	Real	Driver distraction	
[122]	\checkmark		Smartphone and NIR led	Real	Real	Fatigue monitoring, driver distraction	
[127]	\checkmark		Microphone and CAN bus	Real	Real	Driver distraction	
[125]	\checkmark		Smartphone sensors	Real	Real	Driver distraction	
[131]	\checkmark		Visual, auditory, and olfactory stimuli	Real and simulated	Real and simulation	Driver distraction	
[120]	\checkmark		Cameras	Real	Real	Driver distraction	
[132]		\checkmark	Map display device	Real	Real		Route planning
[133]		\checkmark	Smartphone	Real	Real		Route planning
[134]		\checkmark	Smartphone	Real	Real		Tourism guidance
[135]		\checkmark	Map display device	Real	Real		Tourism guidance
[136]		\checkmark	Non specified	Real	Real		Tourism guidance

6.2. Current Challenges

The current challenges of this section are focused mainly on capturing the information associated with the driver's natural perception process and to minimize the degree of distraction to the driver.

(i) Determining the main implicit and explicit characteristics that allow a capturing of the information associated with the driver's natural perception process about the road environment. Actually, the most used explicit characteristics to describe the driver's status are the movement of the eyes, head, hands, and feet, which reflect the state of attention of this and the body's actions of reaction to the traffic's dynamic. These are mostly monitored through vision and infrared cameras installed inside the vehicle, which are susceptible, in many cases, to adverse environmental conditions and poor visibility conditions.

(ii) How to make the sensing of these characteristics as least invasive as possible in order to minimize the degree of distraction to the driver. On the other hand, the implicit characteristics are related to drunkenness, drowsiness, and blood pressure or heart rate, among others, which can be obtained through various types of biomedical sensors attached to the driver. In this case, the installation of these devices can sometimes be inflexible in the face of driving dynamics and even a possible element of distraction and discomfort for the driver himself.

(iii) To address in greater detail the study of the influence that the driver exerts during the driving process on the road environment, an aspect that is evidenced in Table 5 with the reduced number of works that consider the directional interaction $D \rightarrow E$.

(iv) The integrated and complementary approach for the natural context perception module. In this process, the obvious physical limitations of the driver must be considered to analyze aspects related to the calculation of distances and relative speeds between the different vehicles, pedestrians, and objects on the road. It is necessary to establish a holistic approach to develop a comprehensive surrounding environment model from the vision through the sensors and the vision through the driver. This integration to analyze the driving behaviors in the dynamics of the environment will allow a significant increase in the prediction horizon and the precision in the prediction of the driver's intention.

Studies related to the D–E interaction have addressed to a very low degree the correlation between the interpretation of the environment by the driver and the notification to the environment of this, which would undoubtedly modify the situation of the environment itself. The proposed C-ADAS architecture could take advantage of situations of vehicular congestion, unfavorable weather conditions, or dangerous situations of an effective bidirectional interaction between the driver and the environment, which would moderate the behavior of all drivers and consequently mitigate the risks in the environment. In addition to this, there are also the challenges associated with the use of these wearable devices and other on-board sensors and cameras and the way in which they are implemented, so as not to generate distraction or discomfort in the driver, while being able to obtain diverse information with a high predictive value in terms of inferring its future actions. The current state of progress made in the interaction between the driver and the environment, as well as the latent technological challenges, still limit the potential of the proposed C-ADAS.

7. System Evaluation

This section presents aspects related to the evaluation of C-ADAS in the field of road safety, the main mechanisms used, and the evaluation environments, as well as the main evaluation metrics used. We also include some of the challenges associated with the limitations in the use of simulators and evaluation metrics employed in order to meet the design requirements of a C-ADAS proposal, such as the one presented.

7.1. Evaluation Mechanisms

Road traffic represents a complex system made up of multiple independent and interrelated elements. In terms of safety, economy, and fluidity, the evaluation of transport can be associated with the performance of certain main factors, which can be grouped into elements such as the driver, the vehicle, and the road environment. For the most part, the interrelationship between these elements is highly variable and random in nature. The behavior of transport, from the mathematical point of view, can be associated with a stochastic process, where each of its elements can be represented as random variables. An in-depth study of the complexity of this system requires tools that allow for the achieving of a certain degree of reproducibility of the real characteristics present on the roads. To this end, driving simulators have been developed, which are devices used to simulate the driving of a vehicle in an environment with conditions similar to the real characteristics of road traffic. These devices are an effective tool in the training and study of driver behavior, but they also play a very important role in the design and improvement of the HMI elements present in vehicles.

Among the many advantages of using these devices, we can mention the possibility of studying the driver's behavior in situations that may arise while driving and even emulate this behavior in situations that are legally prohibited, such as when driving under the influence of alcohol, or using a cell phone while driving. Another advantage of using simulators corresponds to the reduction of the economic costs associated with carrying out tests in real environments, allowing the generation of high volumes of data from very diverse situations for the training of artificial intelligence models, which guarantees the conditions of reproducibility necessary for statistical analysis of these situations. However, the main limitation is that in these scenarios, drivers are not exposed to a fundamental element: the real risks present on the roads. This can distort the analysis and modeling of

the behavior of drivers in real situations. In the same way, other elements of distraction and influence that are only present in real scenarios are not considered within the simulators, which shape the behaviors of drivers and their reactions to risky situations.

Musa et al. [137] point out that vehicle manufacturers develop and test ADAS technologies following the V-model, which is conventionally used for automotive electronics. The development steps consider the model-in-the-loop and software-in-the-loop tests, while in the validation, the hardware-in-the-loop, vehicle-in-the-loop, and driving tests can be performed. In addition, model predictive control (MPC)-based strategies represent interesting solutions for ADAS evaluation. Vehicle-in-the-loop tests may be useful to evaluate the effective responses of vehicles in safety-critical scenarios and to test it with real-V2X systems, while the combination between hardware-in-the-loop and driving simulators is useful for taking into account human error.

7.2. Performance Metrics

The overall performance of a C-ADAS system must consider not only the precision in estimating the location of remote vehicles, but also the impact of the communication schemes used and the performance of the communication network. The first step in determining this performance is to define a metric that can directly indicate how the security algorithm is performing quantitatively. Generally, for the evaluation of this type of situation, classification metrics are used. Table 6 summarizes the most used metrics in the review of the literature consulted, grouping them fundamentally in regression, classification, and communication performance metrics. The accuracy of hazard detection algorithms to classify situations as dangerous is one such metric [138]. The security algorithm generally runs periodically and at each run-time instance it determines whether a threat exists. The ratio between true positives plus true negatives within the total number of execution instances of the algorithm is defined as the precision. Position-tracking error (PTE), which describes the mean or 95th percentile of the error in tracking a remote vehicle's position, is used to ensure accuracy in position estimation, which is an independent metric of the C-ADAS that is running. Warning detection is performed at the same time as position tracking. The overall performance of C-ADAS depends on the accuracy of the alerts, which depends on the accuracy of tracking the position of the remote vehicles, and this, in turn, depends on the performance of the communications. To analyze this performance, metrics such as the packet error rate (PER) are used, which is the number of packets received incorrectly divided by the total number of packets received. Another metric that can be used to model the robustness of the communications network is the packet loss ratio (PLR), equal to the number of packets not received divided by the total number of packets sent. In this sense, we analyze the evaluation metrics employed in the surveyed works, grouping them into: (i) regression metrics, (ii) classification metrics, and (iii) communication performance metrics. The fulfillment of this task requires the use of evaluation metrics that consider all the elements of the system, focused on the requirements associated with RSA, described mainly in terms of road safety. The robustness of the information is referred to as guaranteeing the reliable delivery of the road safety information required by the RSA (low rates of losses and transmission errors). The freshness of the information meets the delay requirements to guarantee the usefulness of the same in terms of road safety.

Through the analysis shown in Table 6, we can observe as a flaw present in most of the analyzed works, that the consideration, and even more so the evaluation of the behavior of the communications, has not been sufficiently addressed. We assume that this is due to the fact that currently there is a majority trend in the automotive industry to bet on the use of different sensor technologies in vehicles, which are capable of self-supporting the development of the ADAS implemented in these without considering the cooperative operation of these systems between different vehicles and/or with the road infrastructure.

Articles	Regression	Classification	Communication
[82]	90 percentile position error		Packet loss rate
[55,57,58,63]	RMSE position error		
[68]	Total euclidean error sum, horizon euclidean error, and modified Hausdorff distance [139]		
[69]	Mean absolute error		
[59]	Collision risk probability		
[60,70,104,108,116,121,123,131]		Accuracy	
[61]		Accuracy, true positive rate, and true negative rate	
[62]		Recall, accuracy, precision, and F1-score	
[83]	Absolute position error	Accuracy	
[71]	95 percentile of velocity and acceleration error	Accuracy	PER, transmission rate
[65]	95 percentile of absolute position error		
[84]	Spacing position error and velocity tracking error		
[76,77]	95 percent cutoff Euclidean position error		Transmission rate, percentage of packet losses
[78]	90 percentile position tracking error		Transmission rate, PER
[79]	90 percentile position tracking error, average model persistency		Packet length, transmission rate
[81]	Probability of safe breaking		Maximum communication delay, BER
[101]		Accuracy, sensitivity, and specificity	
[102]	Tracking position error		
[106]		Accuracy; premature, timely, and late warning rate	
[109]		false negative rate	
[103]	Tracking position error	Accuracy	
[105]	Driver's behavior volatility	Calinski–Harabasz, silhouette, and Dunn's index	
[119]	Average error of eye-openness	Detection rate	
[129]		Precision, recall, and accuracy	
[126]	Abnormal driver behavior index		
[122]	Normalized mean squared error	Accuracy, precision, recall, and F1-score	
[127]		Accuracy, precision, and recall	
[125]		Precision, recall, and F1-score	
[120]	Mean and standard deviation of yaw angle absolute error	Detection rate, success rate	

Table 6. Classification of the consulted works according to the evaluation metrics used.

7.3. Current Challenges

An important aspect in the evaluation of the C-ADAS systems is the fact of considering their operation and degree of customization as a continuous process. In this sense, systems designed to operate in real time must be capable to adjust dynamically with the new real data. Together with this, it is also important to consider the use of real databases obtained in field measurement campaigns to provide the system with a priori knowledge, which can be

complemented with the processing of data obtained in real time during the tests of system operation. Currently, there is a lack of development of integral evaluation systems in order to be able to evaluate our proposal. Note that, it is impossible for a current simulator to fully replicate real-world driving scenarios, especially from a vehicle and traffic perspective, since there are a set of events and random variables that are difficult to model, such as catastrophic, climatic, environmental, and political-social-cultural events, however, these events have a high incidence on vehicular traffic, the environment, and drivers. Moreover, several of the interactions should be evaluated beforehand, such as the interaction of vehicles to each other as a result of collaborative actions and the communication delays between them. Besides, we consider that it is essential for the effective implementation of the proposed C-ADAS to run as many tests as possible in controlled scenarios, but to make them as real as possible. In the short or medium term, these tests must be executed in hybrid environments where there are vehicles with sensing and communication capabilities and others with the absence of some or none of these capabilities. The latter shows the current relevance of the driving assistance systems, even though there is a growing development in the area of autonomous driving.

8. Conclusions

In this survey, we describe the main points related to the design of a C-ADAS from a holistic and systemic perspective. It is reinforced that the premise of a C-ADAS must take into consideration the three fundamental elements described, namely the driver, vehicle, and environment, and must be designed on an abstraction layer higher than the interaction plane of these elements, which supports the principal module of the C-ADAS system. The main challenges that this industry must set itself must be focused on understanding the current challenges present in each of the three areas of interaction described throughout this work:

- From the point of view of vehicle–environment interaction, we can highlight the standardized incorporation of sensing and communication devices in new vehicles, the reduction in the cost of sensor technologies, and the development of the necessary infrastructure to support the networks of vehicular communication;
- In relation to the vehicle–driver interaction, work should be done to achieve an increase in the degree of customization of the ADAS system, considering the particularities with which each driver acts on the vehicle's driving elements, as well as the way in which it reacts individually to the notices and alerts issued by the system;
- Work must also be done on the design of a flexible HMI interface that adapts continuously to the behavior of the driver in various situations that arise on the road, which is capable of switching between the driver assistance function or taking control of the vehicle before a situation of danger that the driver is not able to resolve favorably;
- One of the areas where greater action is required is in relation to the driver–environment interaction, and the detection of the driver's intention remains today a latent challenge in the design of ADAS systems. To do this, the study of the explicit and implicit characteristics of the driver's behavior must be promoted through the use of devices attached to the interior of the vehicle, which must act in a minimally invasive way so as not to distract or cause general annoyance to the driver, while being able to capture the greatest amount of relevant information from them.

We consider that the necessary future work is the design of a C-ADAS oriented to a higher level of cooperative systems, where cooperation goes beyond the exchange of information provided by the use of communications. We refer to the fact of considering the risk-estimation and decision-making process as a consensual act between different road users. In this approach, the C-ADAS of a vehicle includes, as an additional input in its decision making, the results of the decisions of C-ADAS implemented in neighboring vehicles. The concept of the IoV, a cloud-based service, and the emerging technologies such as the fifth generation of mobile communications can play a crucial role in the sensing and exchange of the information associated with the three main interactions described in this work, namely driver–vehicle, driver–environment, and vehicle–environment, managing the integration of many data sources, which are sometimes very different from each other.

Author Contributions: Conceptualization, S.M.-S., M.F. and S.C.; methodology, S.M.-S. and M.F.; formal analysis, S.M.-S. and M.F.; software, J.F.G.-S.; validation, S.M.-S., M.F. and S.C.; investigation, J.F.G.-S.; resources, J.F.G.-S. and S.M.-S.; writing—original draft preparation, J.F.G.-S.; writing—review and editing, S.M.-S., M.F. and S.C.; Visualization, J.F.G.-S. and S.M.-S.; supervision, S.M.-S. and M.F.; funding acquisition, S.M.-S., J.F.G.-S., M.F. and S.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research was partially supported in Chile by the CONICYT Doctoral Program under Grant No. 21180595, the FONDECYT Iniciación No. 11200659, the FONDECYT Regular No. 1201893, and the Advanced Center for Electrical and Electronic Engineering (AC3E) ANID Basal Project FB0008.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors are grateful for the financial support given by the ANID FONDECYT Iniciación No. 11200659 "Toward High Performance Wireless Connectivity for IoT and Beyond-5G Networks". S.C. has been supported by the Advanced Center for Electrical and Electronic Engineering (AC3E) ANID Basal Project FB0008.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

ABS	Anti-block system
ACC	Adaptive cruise control
ADAS	Advanced driver assistance system
AEB	Autonomous emergency breaking
BNCPD	Bayesian non-negative canonical polyadic decomposition
BSW	Blind-spot warning
C-ADAS	Cooperative advanced driver assistance system
CAN	Controller area network
CARLA	CAR Learning to Act
CAV	Connected and autonomous vehicle
CNN	Convolutional neural network
DSRC	Dedicated short-range communication
EAS	Electronically assisted steering
ECU	Electronic control unit
EEG	Electroencephalogram
ESC	Electronic stability control
ETSI	European Telecommunication Standards Institute
FCW	Forward collision warning
GMM	Gaussian mixture model
GP	Gaussian processes
GPS	Global positioning system
GRU	Gated recurrent unit
HAD	Highly automated driving
H-M	human–machine
HMI	human–machine interface
HMM	Hidden Markov model
IMU	Inertial measurement unit
IoT	Internet of Things
ITS	Intelligent transport system

LCA	Lane change assistance
LDW	Lane departure warning
LED	Light-emission diode
LIDAR	Light detection and ranging
LIN	Local interconnect network
LKA	Lane-keep assistance
LSTM	Long-short-term memory
MPC	Model predictive control
NGSIM	Next generation simulation
NIR	Near infrared
NTSC	National transportation safety council
OBD	On-board diagnostic
OGF	Occupancy grid filtering
PER	Packet error rate
PLR	Packet loss rate
PTE	Packet tracking error
RGB	Red green blue
RMSE	Root mean squared error
RNN	Recurrent neural network
RSA	Road safety applications
RSS	Road safety system
SAE	Society of Automotive Engineering
SDT	Signal detection theory
SPMD	Safety pilot model deployment
SVM	Support vector machine
TTC	Time to collision
V2V	Vehicle-to-vehicle communication
V2X	Vehicle-to-everything communication
VRU	Vulnerable road users

References

- 1. Bohlin, N.L. Safety Belt. U.S. Patent 3,043,625, 10 July 1962.
- ETSI TS 101 539-3; V2X Applications; Part 3: Longitudinal Collision Risk Warning (LCRW) Application Requirements Specification. ETSI: Valbonne, France, 2013; pp. 1–29.
- 3. Rivero, A.Á.L. De sistema mecánico a sistema tecnológico complejo El caso de los automóviles. Contad. Adm. 2014, 59, 11–39.
- Pérez Darquea, D.G. Evolución De Los Dispositivos Electrónicos En Un Automóvil Evolution of Electronic Devices in an Automobile. *INNOVA Res. J.* 2017, 3, 1–7.
- 5. Bosch. Bosch Controller Area Network (CAN) Version 2.0; Bosch: Tokyo, Japan, 1998.
- 6. Cho, K.Y.; Bae, C.H.; Chu, Y.; Suh, M.W. Overview of telematics: A system architecture approach. *Int. J. Automot. Technol.* 2006, 7, 509–517.
- Bylykbashi, K.; Qafzezi, E.; Ikeda, M.; Matsuo, K.; Barolli, L. Fuzzy-based Driver Monitoring System (FDMS): Implementation of two intelligent FDMSs and a testbed for safe driving in VANETs. *Future Gener. Comput. Syst.* 2020, 105, 665–674. [CrossRef]
- 8. Cicchino, J.B. Effectiveness of forward collision warning and autonomous emergency braking systems in reducing front-to-rear crash rates. *Accid. Anal. Prev.* 2017, *99*, 142–152. [CrossRef] [PubMed]
- 9. Cicchino, J.B. Effects of lane departure warning on police-reported crash rates. J. Saf. Res. 2018, 66, 61–70. [CrossRef]
- 10. Cicchino, J.B. Effects of blind spot monitoring systems on police-reported lane-change crashes. *Traffic Inj. Prev.* 2018, 19, 615–622. [CrossRef]
- Bian, Y.; Ding, J.; Hu, M.; Xu, Q.; Wang, J.; Li, K. An Advanced Lane-Keeping Assistance System with Switchable Assistance Modes. *IEEE Trans. Intell. Transp. Syst.* 2020, 21, 385–396. [CrossRef]
- Mehmood, A.; Liaquat, M.; Bhatti, A.I.; Rasool, E. Trajectory Planning and Control for Lane-Change of Autonomous Vehicle. In Proceedings of the 2019 5th International Conference on Control, Automation and Robotics, ICCAR 2019, Beijing, China, 19–22 April 2019; pp. 331–335.
- 13. Flad, M.; Frohlich, L.; Hohmann, S. Cooperative shared control driver assistance systems based on motion primitives and differential games. *IEEE Trans. Hum. Mach. Syst.* 2017, 47, 711–722. [CrossRef]
- 14. Elleuch, I.; Makni, A.; Bouaziz, R. Cooperative Advanced Driver Assistance Systems: A Survey and Recent Trends; Springer: Cham, Switzerland, 2021; pp. 160–170.
- 15. Yue, L.; Abdel-Aty, M.; Wu, Y.; Wang, L. Assessment of the safety benefits of vehicles' advanced driver assistance, connectivity and low level automation systems. *Accid. Anal. Prev.* **2018**, *117*, 55–64. [CrossRef]

- Rasheed, A.; Gillani, S.; Ajmal, S.; Qayyum, A. Vehicular Ad Hoc Network (VANET): A Survey, Challenges, and Applications. In Vehicular Ad-Hoc Networks for Smart Cities; Springer: Singapore, 2017; pp. 39–51.
- 17. Sharma, S.; Kaushik, B. A survey on internet of vehicles: Applications, security issues & solutions. Veh. Commun. 2019, 20, 100182.
- Pathan, K.H.; Patil, M.M. Survey of cooperative advance driver assistance systems. In Proceedings of the 2016 International Conference on Control, Instrumentation, Communication and Computational Technologies (ICCICCT), Kumaracoil, India, 16–17 December 2016; pp. 32–35.
- Hasenjager, M.; Heckmann, M.; Wersing, H. A Survey of Personalization for Advanced Driver Assistance Systems. *IEEE Trans. Intell. Veh.* 2020, 5, 335–344. [CrossRef]
- 20. Marina Martinez, C.; Heucke, M.; Wang, F.Y.; Gao, B.; Cao, D. Driving Style Recognition for Intelligent Vehicle Control and Advanced Driver Assistance: A Survey. *IEEE Trans. Intell. Transp. Syst.* **2018**, *19*, 666–676. [CrossRef]
- Xing, Y.; Lv, C.; Wang, H.; Wang, H.; Ai, Y.; Cao, D.; Velenis, E.; Wang, F.Y. Driver Lane Change Intention Inference for Intelligent Vehicles: Framework, Survey, and Challenges. *IEEE Trans. Veh. Technol.* 2019, *68*, 4377–4390. [CrossRef]
- Bila, C.; Sivrikaya, F.; Khan, M.A.; Albayrak, S. Vehicles of the Future: A Survey of Research on Safety Issues. *IEEE Trans. Intell. Transp. Syst.* 2017, 18, 1046–1065. [CrossRef]
- Siegel, J.E.; Erb, D.C.; Sarma, S.E. A survey of the connected vehicle Landscape—Architectures, enabling technologies, applications, and development areas. *IEEE Trans. Intell. Transp. Syst.* 2018, 19, 2391–2406. [CrossRef]
- 24. Wang, Z.; Wu, Y.; Niu, Q. Multi-Sensor Fusion in Automated Driving: A Survey. IEEE Access 2020, 8, 2847–2868. [CrossRef]
- Martí, E.; De Miguel, M.Á.; García, F.; Pérez, J. A Review of Sensor Technologies for Perception in Automated Driving. *IEEE Intell. Transp. Syst. Mag.* 2019, 11, 94–108. [CrossRef]
- 26. Kaiser, C.; Stocker, A.; Papatheocharous, E. Distracted Driver Monitoring with Smartphones: A Preliminary Literature Review. In Proceedings of the Conference of Open Innovation Association, FRUCT, Tampere, Finland, 12–14 May 2021; pp. 169–176.
- 27. Arumugam, S.; Bhargavi, R. A survey on driving behavior analysis in usage based insurance using big data. *J. Big Data* **2019**, *6*, 86. [CrossRef]
- Zhang, B.; de Winter, J.; Varotto, S.; Happee, R.; Martens, M. Determinants of take-over time from automated driving: A metaanalysis of 129 studies. *Transp. Res. Part F Traffic Psychol. Behav.* 2019, 64, 285–307. [CrossRef]
- Sarker, A.; Shen, H.; Rahman, M.; Chowdhury, M.; Li, F.; Dey, K.; Wang, Y.; Narman, H.S. A review of sensing and communication, human factors, and controller aspects for information-aware connected and automated vehicles. *IEEE Trans. Intell. Transp. Syst.* 2020, 21, 7–29. [CrossRef]
- 30. Smuts, J. Holism and Evolution; Greenwood Press: Westport, CT, USA, 1973.
- Euronews. Segundo Accidente Mortal de un Tesla en "Piloto Automático". 2018. Available online: https://es.euronews.com/20 18/03/31/segundo-accidente-mortal-de-un-tesla-en-piloto-automatico (accessed on 23 January 2021).
- 32. El País. Tesla Reconoce otro Accidente Mortal en un Vehículo que Circulaba con Piloto Automático. 2019. Available online: https://elpais.com/tecnologia/2019/05/17/actualidad/1558075375_210626.html (accessed on 23 January 2021).
- La Vanguardia. Estrella su Tesla en Piloto Automático Contra la Policía Mientras veía una Película. 2020. Available online: https: //lavanguardia.com/motor/actualidad/20200905/33090/estrella-tesla-piloto-automatico-policia-veia-pelicula.html (accessed on 23 January 2021).
- 34. Guo, C.; Kidono, K.; Terashima, R.; Kojima, Y. Humanlike Behavior Generation in Urban Environment Based on Learning-Based Potentials with a Low-Cost Lane Graph. *IEEE Trans. Intell. Veh.* **2018**, *3*, 46–60. [CrossRef]
- 35. *SAE Standard J3016-202104;* Taxonomy and Definitions for Terms Related to Driving Automation Systems for On-Road Motor Vehicles. SAE: Warrendale, PA, USA, 2021.
- Naujoks, F.; Tango, F.; Wolter, S.; Page, Y. L3Pilot—Code of Practice for the development of Automated Driving Functions. L3Pilot Consortium. 2021; pp. 1–171. Available online: https://www.connectedautomateddriving.eu/blog/l3pilot-joint-european-effortboosts-automated-driving/ (accessed on 23 January 2021).
- Wang, Z.; Wu, G.; Barth, M.J.; Bian, Y.; Li, S.E.; Shladover, S.E. A Survey of Multiple Connected and Longitudinal Motion Control on Cooperative Automated Vehicles. *IEEE Intell. Transp. Syst. Mag.* 2020, 50, 965–972. [CrossRef]
- Mardonova, M.; Choi, Y. Review of wearable device technology and its applications to the mining industry. *Energies* 2018, 11, 547. [CrossRef]
- Kashevnik, A.; Shchedrin, R.; Kaiser, C.; Stocker, A. Driver Distraction Detection Methods: A Literature Review and Framework. IEEE Access 2021, 9, 60063–60076. [CrossRef]
- 40. Reina, G.; Johnson, D.; Underwood, J. Radar sensing for intelligent vehicles in urban environments. *Sensors* **2015**, *15*, 14661–14678. [CrossRef]
- Kim, J.; Hong, S.; Baek, J.; Kim, E.; Lee, H. Autonomous vehicle detection system using visible and infrared camera. In Proceedings of the International Conference on Control, Automation and Systems, Jeju, Korea, 17–21 October 2012; pp. 630–634.
- 42. Warren, M.E. Automotive LIDAR Technology. In Proceedings of the 2019 Symposium on VLSI Circuits, Kyoto, Japan, 9–14 June 2019; Volume 1, pp. 254–255.
- Pech, A.H.; Nauth, P.M.; Michalik, R. A new Approach for Pedestrian Detection in Vehicles by Ultrasonic Signal Analysis. In Proceedings of the EUROCON 2019-18th International Conference on Smart Technologies, Novi Sad, Serbia, 1–4 July 2019; pp. 1–5.

- 44. Parate, A.; Ganesan, D. Detecting eating and smoking behaviors using smartwatches. In *Mobile Health: Sensors, Analytic Methods, and Applications*; Springer: Cham, Switzerland, 2017; pp. 175–201.
- Khoa, T.V.A. Wearable Smart Technologies: New Era of Technology. 2015; p. 62. Available online: https://www.theseus.fi/ bitstream/handle/10024/100943/TranVuAnhKhoa_Thesis_FinalVersion.pdf (accessed on 23 January 2021).
- Cadmus-Bertram, L. Using Fitness Trackers in Clinical Research: What Nurse Practitioners Need to Know. J. Nurse Pract. 2017, 13, 34–40. [CrossRef]
- 47. Hanuska, A.; Chandramohan, B.; Bellamy, L.; Burke, P.; Ramanathan, R.; Balakrishnan, V. *Smart Clothing Market Analysis Report*; Berkley University of California: Berkley, CA, USA, 2016; pp. 1–47.
- Kang, J.J.; Venkatraman, S. An Integrated mHealth and Vehicular Sensor Based Alarm System. In Proceedings of the 2018 28th International Telecommunication Networks and Applications Conference (ITNAC), Sydney, Australia, 21–23 November 2018; pp. 1–6.
- Kenney, J.B. Dedicated short-range communications (DSRC) standards in the United States. *Proc. IEEE* 2011, 99, 1162–1182. [CrossRef]
- 50. Zaki, R.W.; Fayed, H.A.; El Aziz, A.A.; Aly, M.H. Outdoor visible light communication in intelligent transportation systems: Impact of snow and rain. *Appl. Sci.* **2019**, *9*, 5453. [CrossRef]
- 51. LTE. Service Requirements for V2X Services; 3GPP TS 22.185 Version 14.3.0 Release 14; ETSI: Valbonne, France, 2017; pp. 1–16.
- Parsons, G. IEEE 802 EC 5G/IMT-2020 Standing Committee Guidelines for IEEE-SA Meetings; IEEE: Piscataway, NJ, USA, 2016; pp. 1–37.
- 53. Neumann, A.; Mytych, M.J.; Wesemann, D.; Wisniewski, L.; Jasperneite, J. Approaches for in-vehicle communication—An analysis and outlook. *Commun. Comput. Inf. Sci.* 2017, 718, 395–411.
- 54. Ji, B.; Zhang, X.; Mumtaz, S.; Han, C.; Li, C.; Wen, H.; Wang, D. Survey on the Internet of Vehicles: Network Architectures and Applications. *IEEE Commun. Stand. Mag.* 2020, *4*, 34–41. [CrossRef]
- Xin, L.; Wang, P.; Chan, C.Y.; Chen, J.; Li, S.E.; Cheng, B. Intention-aware Long Horizon Trajectory Prediction of Surrounding Vehicles using Dual LSTM Networks. In Proceedings of the IEEE Conference on Intelligent Transportation Systems, ITSC, Maui, HI, USA, 4–7 November 2018; pp. 1441–1446.
- 56. USDOT. Next Generation Simulation (NGSIM) Vehicle Trajectories and Supporting Data; USDOT: Topeka, KS, USA, 2017.
- Deo, N.; Trivedi, M.M. Convolutional social pooling for vehicle trajectory prediction. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition Workshops, Salt Lake City, UT, USA, 18–22 June 2018; pp. 1549–1557.
- Deo, N.; Trivedi, M.M. Multi-Modal Trajectory Prediction of Surrounding Vehicles with Maneuver based LSTMs. In Proceedings of the IEEE Intelligent Vehicles Symposium, Changshu, China, 26–30 June 2018; pp. 1179–1184.
- 59. Kim, J.; Kum, D. Collision Risk Assessment Algorithm via Lane-Based Probabilistic Motion Prediction of Surrounding Vehicles. *IEEE Trans. Intell. Transp. Syst.* 2018, 19, 2965–2976. [CrossRef]
- Hou, Y.; Edara, P.; Sun, C. Modeling mandatory lane changing using bayes classifier and decision trees. *IEEE Trans. Intell. Transp.* Syst. 2014, 15, 647–655. [CrossRef]
- Liu, X.; Liang, J.; Xu, B. A Deep Learning Method for Lane Changing Situation Assessment and Decision Making. *IEEE Access* 2019, 7, 133749–133759. [CrossRef]
- 62. Benterki, A.; Boukhnifer, M.; Judalet, V.; Choubeila, M. Prediction of Surrounding Vehicles Lane Change Intention Using Machine Learning. In Proceedings of the 2019 10th IEEE International Conference on Intelligent Data Acquisition and Advanced Computing Systems: Technology and Applications, IDAACS 2019, Metz, France, 18–21 September 2019; Volume 2, pp. 839–843.
- Ding, W.; Shen, S. Online vehicle trajectory prediction using policy anticipation network and optimization-based context reasoning. In Proceedings of the IEEE International Conference on Robotics and Automation, Montreal, QC, Canada, 20–24 May 2019; pp. 9610–9616.
- 64. Dosovitskiy, A.; Ros, G.; Codevilla, F.; Lopez, A.; Koltun, V. CARLA: An Open Urban Driving Simulator. In Proceedings of the 1st Annual Conference on Robot Learning, Mountain View, CA, USA, 13–15 November 2017; pp. 1–16.
- Mahjoub, H.N.; Toghi, B.; Fallah, Y.P. A Driver Behavior Modeling Structure Based on Non-Parametric Bayesian Stochastic Hybrid Architecture. In Proceedings of the IEEE Vehicular Technology Conference, Chicago, IL, USA, 27–30 October 2018; pp. 1–5.
- Batsch, F.; Daneshkhah, A.; Cheah, M.; Kanarachos, S.; Baxendale, A. Performance Boundary Identification for the Evaluation of Automated Vehicles using Gaussian Process Classification. In Proceedings of the 2019 IEEE Intelligent Transportation Systems Conference, ITSC 2019, Auckland, New Zealand, 27–30 October 2019; pp. 419–424.
- 67. Automotive, I. CarMaker Software. 2021. Available online: https://ipg-automotive.com/en/products-solutions/software/ carmaker/ (accessed on 8 October 2021).
- Zyner, A.; Worrall, S.; Nebot, E. Naturalistic Driver Intention and Path Prediction Using Recurrent Neural Networks. *IEEE Trans. Intell. Transp. Syst.* 2019, 21, 1584–1594. [CrossRef]
- Park, S.H.; Kim, B.; Kang, C.M.; Chung, C.C.; Choi, J.W. Sequence-to-Sequence Prediction of Vehicle Trajectory via LSTM Encoder-Decoder Architecture. In Proceedings of the IEEE Intelligent Vehicles Symposium, Changshu, China, 26–30 June 2018; pp. 1672–1678.
- Liu, Y.; Wang, X.; Li, L.; Cheng, S.; Chen, Z. A Novel Lane Change Decision-Making Model of Autonomous Vehicle Based on Support Vector Machine. *IEEE Access* 2019, 7, 26543–26550. [CrossRef]

- Fallah, Y.P. A model-based communication approach for distributed and connected vehicle safety systems. In Proceedings of the 10th Annual International Systems Conference, SysCon 2016, Orlando, FL, USA, 18–21 April 2016; pp. 1–6.
- Kiefer, R.J.; Cassar, M.T.; Flannagan, C.A.; LeBlanc, D.J.; Palmer, M.D.; Deering, R.K.; Shulman, M. Refining the CAMP Crash Alert Timing Approach by Examining "Last-Second" Braking and Lane Change Maneuvers Under Various Kinematic Conditions; National Highway Traffic Safety Administration: Washington, DC, USA, 2003.
- Ahmed-Zaid, F.; Bai, F.; Bai, S.; Basnayake, C.; Bellur, B.; Brovold, S.; Brown, G.; Caminiti, L.; Cunningham, D.; Elzein, H.; et al. Vehicle Safety Communications—Applications (VSC-A) Final Report: Appendix Volume 3 Security; Technical Report No. DOT HS 811 492D; NHTSA: Washington, DC, USA, 2011.
- 74. Ben-Akiva, M.E.; Koutsopoulos, H.N.; Mishalani, R.G.; Yang, Q. Simulation laboratory for evaluating dynamic traffic management systems. J. Transp. Eng. 1997, 123, 283–289. [CrossRef]
- 75. Lynch, N.; Segala, R.; Vaandrager, F.; Weinberg, H.B. Hybrid I/O Automata. Inf. Comput. 2003, 185, 105–157. [CrossRef]
- Huang, C.L.; Fallah, Y.P.; Sengupta, R.; Krishnan, H. Adaptive intervehicle communication control for cooperative safety systems. *IEEE Netw.* 2010, 24, 6–13. [CrossRef]
- 77. Huang, C.L.; Fallah, Y.P.; Sengupta, R.; Krishnan, H. Intervehicle transmission rate control for cooperative active safety system. *IEEE Trans. Intell. Transp. Syst.* 2011, 12, 645–658. [CrossRef]
- Mahjoub, H.N.; Toghi, B.; Osman Gani, S.M.; Fallah, Y.P. V2X system architecture utilizing hybrid Gaussian process-based model structures. In Proceedings of the 13th Annual IEEE International Systems Conference, Proceedings, SysCon 2019, Orlando, FL, USA, 8–11 April 2019; pp. 1–7.
- Mahjoub, H.N.; Raftari, A.; Valiente, R.; Fallah, Y.P.; Mahmud, S.K. Representing Realistic Human Driver Behaviors using a Finite Size Gaussian Process Kernel Bank. In Proceedings of the 2019 IEEE Vehicular Networking Conference (VNC), Los Angeles, CA, USA, 4–6 December 2019.
- 80. USDOT. Safety Pilot Model Deployment Data; USDOT: Topeka, KS, USA, 2019.
- Vinel, A.; Lyamin, N.; Isachenkov, P. Modeling of V2V Communications for C-ITS Safety Applications: A CPS Perspective. *IEEE Commun. Lett.* 2018, 22, 1600–1603. [CrossRef]
- 82. Mahjoub, H.N.; Tahmasbi-Sarvestani, A.; Kazemi, H.; Fallah, Y.P. A Learning-Based Framework for Two-Dimensional Vehicle Maneuver Prediction over V2V Networks. In Proceedings of the 2017 IEEE 15th International Conference on Dependable, Autonomic and Secure Computing, 2017 IEEE 15th International Conference on Pervasive Intelligence and Computing, 2017 IEEE 3rd International Conference on Big Data Intelligence and Computing and Cyber Science and Technology Congress, Orlando, FL, USA, 6–10 November 2018; pp. 156–163.
- Du, H.; Leng, S.; Wu, F.; Chen, X.; Mao, S. A New Vehicular Fog Computing Architecture for Cooperative Sensing of Autonomous Driving. *IEEE Access* 2020, *8*, 10997–11006. [CrossRef]
- 84. Moradi-Pari, E.; Mahjoub, H.N.; Kazemi, H.; Fallah, Y.P.; Tahmasbi-Sarvestani, A. Utilizing Model-Based Communication and Control for Cooperative Automated Vehicle Applications. *IEEE Trans. Intell. Veh.* **2017**, *2*, 38–51. [CrossRef]
- United States Environmental Protection Agency. 2021. Available online: https://www.epa.gov/transportation-air-pollutionand-climate-change (accessed on 20 March 2021).
- Zhou, H.; Laval, J.; Zhou, A.; Wang, Y.; Wu, W.; Qing, Z.; Peeta, S. Review of Learning-Based Longitudinal Motion Planning for Autonomous Vehicles: Research Gaps Between Self-Driving and Traffic Congestion. *Transp. Res. Rec.* 2022, 2676, 324–341. [CrossRef]
- Xu, B.; Ban, X.J.; Bian, Y.; Li, W.; Wang, J.; Li, S.E.; Li, K. Cooperative Method of Traffic Signal Optimization and Speed Control of Connected Vehicles at Isolated Intersections. *IEEE Trans. Intell. Transp. Syst.* 2019, 20, 1390–1403. [CrossRef]
- Zheng, Y.; Wang, J.; Li, K. Smoothing Traffic Flow via Control of Autonomous Vehicles. *IEEE Internet Things J.* 2020, 7, 3882–3896. [CrossRef]
- Wang, J.; Gong, S.; Peeta, S.; Lu, L. A real-time deployable model predictive control-based cooperative platooning approach for connected and autonomous vehicles. *Transp. Res. Part Methodol.* 2019, 128, 271–301. [CrossRef]
- Zhou, A.; Gong, S.; Wang, C.; Peeta, S. Smooth-Switching Control-Based Cooperative Adaptive Cruise Control by Considering Dynamic Information Flow Topology. *Transp. Res. Rec.* 2020, 2674, 444–458. [CrossRef]
- 91. Zhou, A.; Peeta, S.; Yang, M.; Wang, J. Cooperative signal-free intersection control using virtual platooning and traffic flow regulation. *Transp. Res. Part Emerg. Technol.* **2022**, 138, 103610. [CrossRef]
- 92. Wang, J.; Peeta, S.; Lu, L.; Li, T. Multiclass information flow propagation control under vehicle-to-vehicle communication environments. *TRansportation Res. Part B Methodol.* **2019**, *129*, 96–121. [CrossRef]
- 93. Wang, J.; Peeta, S.; He, X. Multiclass traffic assignment model for mixed traffic flow of human-driven vehicles and connected and autonomous vehicles. *Transp. Res. Part Methodol.* **2019**, *126*, 139–168. [CrossRef]
- 94. Wang, J.; Lu, L.; Peeta, S.; He, Z. Optimal toll design problems under mixed traffic flow of human-driven vehicles and connected and autonomous vehicles. *Transp. Res. Part C Emerg. Technol.* **2021**, *125*, 102952. [CrossRef]
- 95. Gandhi, T.; Trivedi, M.M. Vehicle Surround Capture: Survey of Techniques and a Novel Omni-Video-Based Approach for Dynamic Panoramic Surround Maps. *IEEE Trans. Intell. Transp. Syst.* 2006, 7, 293–308. [CrossRef]
- 96. Palazzi, A.; Abati, D.; Calderara, S.; Solera, F.; Cucchiara, R. Predicting the Driver's Focus of Attention: The DR (eye) VE Project. *IEEE Trans. Pattern Anal. Mach. Intell.* **2018**, *41*, 1720–1733. [CrossRef] [PubMed]

- Demmel, S.; Gruyer, D.; Rakotonirainy, A. V2V/V2I augmented maps: State-of-the-art and contribution to real-time crash risk assessment Résumé. In Proceedings of the 20th Canadian Multidisciplinary Road Safety Conference, Niagara Falls, ON, USA, 6–9 June 2010.
- 98. Katsaros, K.; Catapult, D.; Dianati, M. A Conceptual 5G Vehicular Networking Architecture the driving force to connected autonomy. In *Key Enabling Technologies for 5G Mobile Communications. Electronic Engineering*; Springer: Berlin, Germany, 2017.
- Nkenyereye, L.; Jang, J.w. Integration of Big Data for Connected Cars Applications Based on Tethered Connectivity. Procedia-Procedia Comput. Sci. 2016, 98, 554–559. [CrossRef]
- 100. Woo, S.; Jo, H.J.; Lee, D.H. A Practical Wireless Attack on the Connected Car and Security Protocol for In-Vehicle CAN. *IEEE Trans. Intell. Transp. Syst.* 2015, *16*, 993–1006. [CrossRef]
- 101. Wang, W.; Xi, J.; Zhao, D. Learning and Inferring a Driver 's Braking Action in Car-Following Scenarios. *IEEE Trans. Veh. Technol.* **2018**, *67*, 1–13.
- Dang, R.; Wang, J.; Li, S.E.; Li, K. Coordinated Adaptive Cruise Control System With Lane-Change Assistance. *IEEE Trans. Intell. Transp. Syst.* 2015, 16, 2373–2383. [CrossRef]
- Zhu, B.; Yan, S.; Zhao, J.; Deng, W. Personalized Lane-Change Assistance System with Driver Behavior Identification. *IEEE Trans. Veh. Technol.* 2018, 67, 10293–10306. [CrossRef]
- Su, C.; Deng, W.; Sun, H.; Wu, J.; Sun, B.; Yang, S. Forward collision avoidance systems considering drivers driving behavior recognized by Gaussian Mixture Model. *IEEE Intell. Veh. Symp.* 2017, 535–540.
- Mantouka, E.G.; Barmpounakis, E.N.; Vlahogianni, E.I. Identifying driving safety profiles from smartphone data using unsupervised learning. *Saf. Sci.* 2019, 119, 84–90. [CrossRef]
- 106. Yang, W.; Wan, B.; Qu, X. A Forward Collision Warning System Using Driving Intention Recognition of the Front Vehicle and V2V Communication. *IEEE Access* 2020, *8*, 11268–11278. [CrossRef]
- Bavendiek, J.; Oliveira, E.; Eckstein, L. A Novel Method for Designing Metaphor-Based Driver-Vehicle Interaction Concepts in Automated Vehicles. In *International Conference on Applied Human Factors and Ergonomics*; Springer: Cham, Switzerland, 2019; Volume 964, pp. 15–26.
- Iranmanesh, S.M.; Nourkhiz Mahjoub, H.; Kazemi, H.; Fallah, Y.P. An adaptive forward collision warning framework design based on driver distraction. *IEEE Trans. Intell. Transp. Syst.* 2018, 19, 3925–3934. [CrossRef]
- 109. Sun, Q.; Zhang, H.; Li, Z.; Wang, C.; Du, K. ADAS Acceptability improvement based on self-learning of individual driving characteristics: A case study of lane change warning system. *IEEE Access* 2019, 7, 81370–81381. [CrossRef]
- Choi, J.K.; Kim, K.; Kim, D.; Choi, H.; Jang, B. Driver-adaptive vehicle interaction system for the advanced digital cockpit. In Proceedings of the International Conference on Advanced Communication Technology, ICACT, Chuncheon, Korea, 11–14 February 2018; pp. 307–310.
- 111. Dargahi Nobari, K.; Albers, F.; Bartsch, K.; Bertram, T. Online Feedback Control for Driver-Vehicle Interaction in Automated Driving. In *Advances in Human Aspects of Transportation*; Springer: Cham, Switzerland, 2020; Volume 1212, pp. 159–165.
- 112. Yang, X.; Liu, L.; Vaidya, N.H.; Zhao, F. A Vehicle-to-Vehicle Communication Protocol. In Proceedings of the The First Annual International Conference on Mobile and Ubiquitous Systems: Networking and Services, Boston, MA, USA, 26 August 2004.
- Lv, C.; Wang, H.; Cao, D.; Zhao, Y.; Sullman, M.; Auger, D.J.; Brighton, J.; Matthias, R.; Skrypchuk, L.; Mouzakitis, A. A Novel Control Framework of Haptic Take-Over System for Automated Vehicles. In Proceedings of the IEEE Intelligent Vehicles Symposium, Changshu, China, 26–30 June 2018; pp. 1596–1601.
- 114. Wan, J.; Wu, C. The effects of lead time of take-over request and non-driving tasks on taking-over control of automated vehicles. *IEEE Trans.-Hum.-Mach. Syst.* **2018**, *48*, 582–591. [CrossRef]
- Li, Y.; Lv, C.; Xue, J. A novel predictive haptic control interface for automation-to-human takeover of automated vehicles. In Proceedings of the IEEE Intelligent Vehicles Symposium, Paris, France, 9–12 June 2019; pp. 994–999.
- 116. Qiao, Y.; Zeng, K.; Xu, L.; Yin, X. A smartphone-based driver fatigue detection using fusion of multiple real-time facial features. In Proceedings of the 2016 13th IEEE Annual Consumer Communications and Networking Conference, CCNC 2016, Las Vegas, NV, USA, 9–12 January 2016; pp. 230–235.
- 117. AL-Anizy, G.J.; Nordin, M.J.; Razooq, M.M. Automatic Driver drowsiness detection using Haar Algorithm and Support Vector Machine Techniques. *Asian J. Appl. Sci.* 2015, *8*, 149–157. [CrossRef]
- 118. Baştan, M.; Bukhari, S.S.; Breuel, T. Active Canny: Edge detection and recovery with open active contour models. *IET Image Process* **2017**, *11*, 1325–1332. [CrossRef]
- 119. Mandal, B.; Li, L.; Wang, G.S.; Lin, J. Towards Detection of Bus Driver Fatigue Based on Robust Visual Analysis of Eye State. *IEEE Trans. Intell. Transp. Syst.* 2017, *18*, 545–557. [CrossRef]
- 120. Yuen, K.; Trivedi, M.M. An occluded stacked hourglass approach to facial landmark localization and occlusion estimation. *IEEE Trans. Intell. Veh.* **2018**, *2*, 321–331. [CrossRef]
- 121. Eraqi, H.M.; Abouelnaga, Y.; Saad, M.H.; Moustafa, M.N. Driver distraction identification with an ensemble of convolutional neural networks. *J. Adv. Transp.* **2019**, 2019, 4125865. [CrossRef]
- 122. Janveja, I.; Nambi, A.; Bannur, S.; Gupta, S.; Padmanabhan, V. InSight: Monitoring the State of the Driver in Low-Light Using Smartphones. In Proceedings of the ACM on Interactive, Mobile, Wearable and Ubiquitous Technologies, New York, NY, USA, 18 September 2020; Volume 4, pp. 1–29. [CrossRef]

- Kapoor, K.; Pamula, R.; Murthy, S.V. Real-Time Driver Distraction Detection System Using Convolutional Neural Networks. In Proceedings of ICETIT 2019; Springer International Publishing: Cham, Switzerland, 2020; Volume 605, pp. 280–291.
- 124. Kaggle. State Farm Distracted Driver Detection; Kaggle: San Francisco, CA, USA, 2019.
- 125. Xie, J.; Hilal, A.R.; Kulic, D. Driver Distraction Recognition Based on Smartphone Sensor Data. In Proceedings of the 2018 IEEE International Conference on Systems, Man, and Cybernetics, SMC 2018, Miyazaki, Japan, 7–10 October 2018; pp. 801–806.
- Hu, J.; Xu, L.; He, X.; Meng, W. Abnormal Driving Detection Based on Normalized Driving Behavior. *IEEE Trans. Veh. Technol.* 2017, 66, 6645–6652. [CrossRef]
- 127. Qi, B.; Zhao, W.; Wang, X.; Li, S.; Runge, T. A low-cost driver and passenger activity detection system based on deep learning and multiple sensor fusion. In Proceedings of the ICTIS 2019—5th International Conference on Transportation Information and Safety, Liverpool, UK, 14–17 July 2019; pp. 170–176.
- 128. CARB. OBD II Regulations and Rulemaking; CARB: El Monte, CA, USA, 2021.
- 129. Rohit, F.; Kulathumani, V.; Kavi, R.; Elwarfalli, I.; Kecojevic, V.; Nimbarte, A. Real-time drowsiness detection using wearable, lightweight brain sensing headbands. *IET Intell. Transp. Syst.* 2017, *11*, 255–263. [CrossRef]
- Li, G.; Chung, W.Y. Combined EEG-gyroscope-TDCS brain machine interface system for early management of driver drowsiness. IEEE Trans. Hum. Mach. Syst. 2018, 48, 50–62. [CrossRef]
- Guo, Y.; Wang, X.; Yuan, Q.; Liu, S.; Liu, S. Transition characteristics of driver's intentions triggered by emotional evolution in two-lane urban roads. *Intell. Transp. Syst.* 2020, 14, 1788–1798. [CrossRef]
- 132. Xu, W.; Gao, R. Prediction of Road Conditions Ahead based on Travel Plans. In Proceedings of the 2019 IEEE 5th International Conference on Computer and Communications, ICCC 2019, Chengdu, China, 6–9 December 2019; pp. 262–266.
- 133. Withanage, P.; Liyanage, T.; Deeyakaduwe, N.; Dias, E.; Thelijjagoda, S. Voice-based Road Navigation System Using Natural Language Processing (NLP). In Proceedings of the 2018 IEEE 9th International Conference on Information and Automation for Sustainability, ICIAfS 2018, Colombo, Sri Lanka, 21–24 December 2018.
- Keerthana, R.; Kumar, T.A.; Manjubala, P.; Pavithra, M. An Interactive Voice Assistant System for Guiding the Tourists in Historical places. In Proceedings of the 2020 International Conference on System, Computation, Automation and Networking, ICSCAN 2020, Pondicherry, India, 3–4 July 2020.
- Zhou, X.; Sun, B.; Li, S.; Liu, S. Tour Route Planning Algorithm Based on Precise Interested Tourist Sight Data Mining. *IEEE Access* 2020, *8*, 153134–153168. [CrossRef]
- 136. Rathnayake, W.G. Google Maps Based Travel Planning Analyzing System (TPAS). In Proceedings of the 2018 International Conference on Current Trends towards Converging Technologies, ICCTCT 2018, Coimbatore, India, 1–3 March 2018; pp. 1–5.
- 137. Musa, A.; Pipicelli, M.; Spano, M.; Tufano, F.; De Nola, F.; Di Blasio, G.; Gimelli, A.; Misul, D.A.; Toscano, G. A review of model predictive controls applied to advanced driver-assistance systems. *Energies* **2021**, *14*, 7974. [CrossRef]
- Fallah, Y.P.; Khandani, M.K. Context and Network Aware Communication Strategies for Connected Vehicle Safety Applications. IEEE Intell. Transp. Syst. Mag. 2016, 8, 92–101. [CrossRef]
- 139. Dubuisson, M.P.; Jain, A.K. A modified Hausdorff distance for object matching. In Proceedings of the International Conference on Pattern Recognition, Jerusalem, Israel, 9–13 October 1994; Volume 1.